

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Kanne, Stefan; Korn, Olaf; Uhrig-Homburg, Marliese

# Working Paper Stock Illiquidity, option prices, and option returns

CFR Working Paper, No. 16-08

**Provided in Cooperation with:** Centre for Financial Research (CFR), University of Cologne

*Suggested Citation:* Kanne, Stefan; Korn, Olaf; Uhrig-Homburg, Marliese (2016) : Stock Illiquidity, option prices, and option returns, CFR Working Paper, No. 16-08, University of Cologne, Centre for Financial Research (CFR), Cologne

This Version is available at: https://hdl.handle.net/10419/146777

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# **Stock Illiquidity, Option Prices, and Option Returns**

Stefan Kanne<sup>\*</sup>, Olaf Korn<sup>\*\*</sup>, and Marliese Uhrig-Homburg<sup>\*\*\*</sup>

Current version: July 2016

## Abstract

We provide evidence of a strong effect of the underlying stock's illiquidity on option prices by showing that the average absolute difference between historical and implied volatility increases with stock illiquidity. This pattern translates into significant excess returns of option trading strategies that are not explained by common risk factors. Simulation results show, however, that our results can be explained by the hedging costs of market makers who are net long in options on some underlyings and net short in options on other underlyings. Our empirical findings are robust with respect to the chosen illiquidity measure, the measure of option expensiveness, and the return period.

JEL Classification: G12; G13

Keywords: Illiquidity, equity options, option returns, option strategies

We are grateful for helpful comments from Turan Bali, Jay Cao, Bing Han, Jeff Hobbs, Ruslan Goyenko, Alexander Kempf, Philipp Schuster, Gustavo Schwenkler, and participants at the 2015 OptionMetrics Research Conference, the 2015 Meeting of the German Finance Association (DGF), the 2016 Meeting of the Swiss Society for Financial Market Research (SGF), the 2016 Meeting of the Eastern Finance Association, and the research seminar at the Centre for Financial Research Cologne (CFR). This work was supported by the Deutsche Forschungsgemeinschaft [UH 107/4-1, KO 2285/3-1].

\* Stefan Kanne, Chair of Financial Engineering and Derivatives, Karlsruhe Institute of Technology (KIT), P.O. Box 6980, D-76049 Karlsruhe, Germany; email sckanne@gmail.com.

\*\* Olaf Korn, Chair of Finance, Georg-August-Universität Göttingen and Centre for Financial Research Cologne (CFR), Platz der Göttinger Sieben 3, D-37073 Göttingen, Germany; phone +49 551 39 7265; fax +49 551 39 7665; email okorn@uni-goettingen.de.

\*\*\* Marliese Uhrig-Homburg, Chair of Financial Engineering and Derivatives, Karlsruhe Institute of Technology (KIT), P.O. Box 6980, D-76049 Karlsruhe, Germany; phone +49 721 608 48183; fax +49 721 608 48190; email uhrig@kit.edu.

In a Black–Scholes (1973) economy, intermediaries can perfectly hedge their options positions via dynamic trading strategies in the underlying and a risk-free bond. In reality, perfect hedging is infeasible or too costly due to market incompleteness and market frictions. Demand-based option pricing theory addresses this issue by showing how market makers account for unhedgeable risks depending on the sign and magnitude of the net demand they face. If there is higher end user demand to buy a specific option series than to sell it, market makers will charge a higher option price as compensation for risks to be taken. Conversely, if market makers face end user selling pressure, they will lower the option price. Empirical evidence by Bollen and Whaley (2004), Gârleanu, Pedersen, and Poteshman (2009), and Muravyev (2016) shows that demand pressure indeed influences option prices in this way.

Taken together, the demand-based option pricing literature recognizes that option prices are influenced by the sign and magnitude of market makers net option position and the unhedgeable risk inherent to market makers' option inventories. Unhedgeable risks include the inability to continuously rebalance hedging positions (Gârleanu, Pedersen, and Poteshman, 2009). Naturally, the liquidity of the underlying affects hedging costs and could also decrease rebalancing frequencies and thus increase inventory risk. Stock illiquidity should therefore affect the way market makers' inventories impact option prices, depending on whether these inventories are positive or negative. Surprisingly, however, very little is known about the connection between stock illiquidity and option prices and the scarce empirical evidence (Karakaya, 2014; Christoffersen et al., 2015; Choy and Wei, 2016) is mixed. Our paper documents a strong relation between stock illiquidity and option prices. In our cross-sectional analysis, we find that the absolute deviations of implied from historical volatility increase with stock illiquidity. Such a relation naturally arises if market makers are net long in options on some stocks and net short in others, which is a realistic scenario. Although there is evidence that market makers are, on average, net long in options written on individual stocks (Lakonishok et al., 2007; Ni, Pan, and Poteshman, 2008; Gârleanu, Pedersen, and Poteshman, 2009; Muravyev, 2016), the standard deviation is very large (Ni, Pan, and Poteshman, 2008; Muravyev, 2016), which implies that we find both net long and net short positions of market makers, depending on the particular option series.

The idea that stock illiquidity is an important driver of option prices is further tested by looking at option trading strategies that build on information on the underlying stock's illiquidity. These

strategies deliver significant excess returns that cannot be explained by standard risk factors. Our results also show that a large part of the returns of option trading strategies based on the difference between historical and implied volatility (Goyal and Saretto, 2009) can be captured by stock illiquidity.

Our empirical investigation proceeds in four steps. First, we study the relation between option expensiveness and stock illiquidity. In our base case, we measure option expensiveness as the difference between option implied volatility and realized historical volatility and relate this difference to the underlying stock's Amihud (2002) illiquidity measure. Second, we investigate how greater differences between historical and implied volatility translate into higher option excess returns. If higher volatility differences are not due to the market's superior volatility forecasting abilities but are the result of market imperfections, they should predict higher option excess returns. We use trading strategies with straddles and delta-hedged options throughout the paper to obtain option excess returns. Third, we investigate different explanations for the observed patterns of option prices, option returns, and stock illiquidity. A first test investigates whether option returns can be explained by standard risk factors suggested in the literature. A second test uses a simulation study to see if the magnitude of our empirical findings is consistent with market makers accounting for transaction costs in the underlying stocks and being net long in options on some underlyings and net short in options on other underlyings. In the fourth and final step of our analysis, we perform different robustness checks with respect to the chosen illiquidity measure, the measure of option expensiveness, and the return period.

Our paper adds to a small but growing literature that attempts to relate market frictions and option returns. Christoffersen et al. (2015) investigate how option illiquidity and stock illiquidity affect delta-hedged option returns. They document significant premiums for illiquid options but do not find clear evidence for the role of stock illiquidity. Choy and Wei (2016) find premiums for options' illiquidity risk, however, option returns do not significantly load on a stock market liquidity factor. Although market makers are, on average, long in individual equity options, they could be short, especially in options with highly illiquid underlyings, which is likely to conceal a direct connection between stock illiquidity and option returns. Cao and Han (2013) study the effects of systematic and idiosyncratic volatility on option returns and show that options on high idiosyncratic volatility stocks have lower returns than options on low idiosyncratic risk stocks. They have in mind a setting where speculative investors buy options on stocks with high

idiosyncratic volatility. These speculative investors, demanding liquidity in the option market, are willing to pay a premium, while the market makers who are net short find it costly to provide these options and charge a higher price. Frazzini and Pedersen (2012) advocate the role of embedded leverage in alleviating investors' leverage constraints. They provide evidence that intermediaries who meet investors' demand for equity options with higher embedded leverage are compensated for their higher risk. Goyal and Saretto (2009) find that long–short option portfolios based on the deviation between historical and implied volatility produce excess returns that cannot be explained by standard risk factors.<sup>1</sup> Karakaya (2014) suggests that market frictions could help explain the returns of Goyal and Saretto's (2009) portfolios and shows that the premium earned by the strategy depends on overall market and funding liquidity. Our study can be seen as a cross-sectional version of this test, because we show that the trading profits from Goyal and Saretto's (2009) strategy are much higher for options on less liquid stocks.

Finally, our findings complement the literature on the impact of stock illiquidity on option illiquidity. Two recent examples are the studies of Engle and Neri (2010) and Goyenko, Ornthanalai, and Tang (2015), who show empirically that both the transaction costs of the initial option hedge as well as the costs of rebalancing the hedge position widen an option's bid–ask spread. We provide empirical evidence that hedging costs influence not only bid and ask prices (and therefore the bid–ask spread) but also the mid price of options.

The remainder of the paper is organized as follows. Section I provides background on the data used in our empirical study. Section II presents our main results on the relations between option prices (returns) and the underlying stock's illiquidity. Section III investigates different explanations for the observed patterns. Section IV presents a robustness analysis and Section V concludes the paper.

<sup>&</sup>lt;sup>1</sup> Goyal and Saretto (2009) advocate a behavioral explanation based on overreaction. Karakaya's (2014) findings on put portfolios and the results of An et al. (2014) on the effects of earnings dispersion, however, raise doubts that overreaction is the only reason.

## I. Data Set and Data Processing

#### A. Data Sources and Filters

Our primary data source is the OptionMetrics Ivy DB database. This database contains information on all U.S. exchange-listed individual equity options, including daily closing bid and ask quotes, trading volumes, open interest, and options' Greeks (delta, gamma, vega) and implied volatility. The delta and implied volatility we use are calculated by OptionMetrics' proprietary algorithms that account for discrete dividend payments and the early exercise of American options.<sup>2</sup> The database also contains the closing prices, trading volumes, and information on dividend payments, stock splits, and total return calculations for the options' underlying stocks. Our Ivy DB database sample period is from January 1996 to August 2015.

We use similar filters as in previous studies (Goyal and Saretto, 2009; Cao and Han, 2013; Karakaya, 2014) to minimize the impact of recording errors. We drop all observations where the option bid price is zero and the bid price is higher than the ask price. In addition, we eliminate options with a bid–ask spread smaller than the minimum tick size (\$0.05 for options trading below \$3 and \$0.1 for all other options). We remove observations with zero open interest and require a non-missing delta and implied volatility to keep the observation in the sample. Options with an ex dividend date during the holding period are excluded. We also eliminate option observations that violate obvious no arbitrage conditions such as  $S \ge C \ge \max(S - Ke^{-rT}, 0)$  for call price *C*, underlying stock price *S*, strike *K*, risk-free rate *r*, and time to maturity *T*.

## B. Return Calculations

Our portfolio formation follows that of Goyal and Saretto (2009). To reduce the impact of stock price risk on an option's return, we use two kinds of portfolios. The first contains delta-hedged call options and the second consists of straddles. The portfolios of options and their underlying stocks are based on information available on the first trading day (usually a Monday) after the expiration day of the month.<sup>3</sup> We consider only options that mature the next month and restrict our sample to at-the-money (ATM) options with moneyness (defined as the ratio of the strike price to the stock price) between 0.975 and 1.025 on the day of portfolio formation (usually a

<sup>&</sup>lt;sup>2</sup> We refer the reader to the Ivy DB reference manual for further details.

<sup>&</sup>lt;sup>3</sup> Before February 2015, all options expire on the Saturday following the third Friday of the month. Thereafter, they expire at the close of business of the expiration month's third Friday.

Monday). Throughout the sample period, we have 156,849 straddle pairs of calls and puts and 197,374 delta-hedged call observations. To avoid microstructure biases, we follow Goyal and Saretto (2009) and start trading the trading day (usually a Tuesday) after the day on which we obtain the trading signal (usually a Monday) and hold the option until maturity. This implies that the option payoffs and the returns of stock positions used for delta hedging are based on the last closing stock prices prior to expiration.

#### B.1. Delta-Hedged Option Returns

We calculate the returns of delta-hedged call options portfolios that buy one option contract and sell delta shares of the underlying stock, with the net investment earning the risk-free rate (obtained from Kenneth French's data library). The return is calculated as

$$\Pi_{t,t+\tau} = \frac{\max(S_{t+\tau} - K, 0) - \Delta_{C,t} S_{t+\tau} - (C_t - \Delta_{C,t} S_t) e^{r\tau}}{Abs(C_t - \Delta_{C,t} S_t)},$$
(1)

where *K* is the option's strike price,  $\Delta_{C,t}$  is the option's delta, and *C* and *S* are the mid prices of the call and the underlying stock, respectively, at *t*, the trading initiation date, and  $t + \tau$ , the last trading day prior to expiration. We scale the dollar return by the absolute value of the option bought and the delta shares ( $\Delta_{C,t} S_t$ ) sold at trading initiation.

## B.2. Straddle Returns

Straddles are formed as a combination of one call and one put on the same underlying with identical strike prices and maturity. Although we restrict our sample to options with moneyness between 0.975 and 1.025 and then choose the call and put closest to being ATM for each month and each underlying, there could be a difference between the call and put strikes. The straddle returns are therefore calculated as

$$\Pi_{t,t+\tau} = \frac{\max(S_{t+\tau} - K_{Call}, 0) + \max(K_{Put} - S_{t+\tau}, 0) - (C_t + P_t)e^{r\tau}}{C_t + P_t},$$
(2)

where  $K_{Call}$  and  $K_{Put}$  could be slightly different.

#### C. Measures of Option Expensiveness

HV-IV: Our main measure of option expensiveness is the difference between a benchmark estimate of volatility from historical stock return data (HV) and the option's implied volatility (IV). The lower the measure, the more expensive the option. HV–IV is a standard measure for investigating the impact of frictions on option prices and, for example, the empirical work by Bollen and Whaley (2004) and Gârleanu, Pedersen, and Poteshman (2009) concentrates on the impact of demand pressure on this form of expensiveness. As the implied volatility for one stock on the portfolio formation date (t - 1), we use the average of the implied volatilities of the respective put and call options on the stock. For the delta-hedge call strategy, this volatility is replaced by the implied volatility of the call option. The historical volatility is, following Goyal and Saretto (2009), the standard deviation of daily stock returns using the 12 months preceding portfolio formation, unless stated otherwise.

*PVOL–CVOL:* HV–IV is a measure of the average expensiveness of put and call options because IV uses the average implied volatility of puts and calls. However, stock illiquidity and demand pressure could also have an impact on the relative pricing of puts and calls. In accordance with this idea of order imbalances in put and call options, Bali and Hovakimian (2009) have shown that the difference between put and call implied volatility can predict future stock returns. Therefore, we use the difference between the implied volatilities of ATM put options (PVOL) and ATM call options (CVOL) as a measure of the relative expensiveness of put options compared to calls.

#### D. Measures of Stock Illiquidity

Our main measure of underlying stock illiquidity is the average of the daily Amihud (2002) measure over the month preceding the portfolio formation date. Goyenko, Holden, and Trzcinka (2009) show that the Amihud measure is the best low-frequency market impact measure and also a good proxy for effective and realized bid–ask spreads. We also use Roll's (1984) and Corwin and Schultz's (2012) stock spread estimates as well as the stock's trading volume and market capitalization in our robustness checks. Details on the liquidity measure calculations can be found in Appendix A.

## **II. Main Results**

#### A. Stock Illiquidity and Option Expensiveness

Our first analysis examines the relation between option expensiveness and stock illiquidity. Every month, on the portfolio formation date, we first sort stocks into quintiles based on their Amihud measure; then the stocks in each Amihud quintile are sorted into quintiles based on HV–IV. For every month throughout the observation period, we calculate the mean HV–IV for each combination of Amihud quintiles and the HV–IV quintiles. Table I reports the time-series averages and t-statistics of these monthly means. In addition, the last two columns show the time-series average of the mean and standard deviation of HV–IV within the Amihud quintiles.

If stock illiquidity affects option prices, we expect the distribution of HV–IV to change with illiquidity. If market makers were only buyers of individual equity options, the mean HV–IV should increase with illiquidity and, if market makers were only sellers, the mean HV–IV should decrease. The last two columns in Table I show that the mean HV–IV differs only insignificantly between the Amihud quintiles, while the standard deviation doubles from the lowest to the highest quintile, with a t-statistic above 10 for the difference between the lowest and highest quintiles. In principle, an increase of the HV–IV standard deviation is in line with market makers being net long in options on some stocks and net short in options on others. Moreover, the double-sorted portfolios show smooth monotonic increases and decreases of the average HV–IV values with illiquidity. The relatively "expensive" options located in the first and second HV–IV quintile columns show a monotonic decrease of HV–IV with higher illiquidity, while the "cheap" ones in the two highest HV–IV quintile columns show a monotonic increase. The difference between the highest and lowest Amihud quintiles for the center HV–IV quintile column is the only one that is not highly significant.

## [Insert Table I about here]

These observations suggest that higher illiquidity leads to broader dispersion of implied volatility around historical volatility, with cheap options becoming cheaper and expensive options becoming more expensive with increasing stock illiquidity.<sup>4</sup> The results in Table I are consistent with market makers pricing options in a way that takes hedging costs due to stock illiquidity into account while being net long in options on some stocks and net short in options on others. To check whether observed end user net demand is in line with this argument, we use a publicly available net option demand data sample of all closing short and long open interests on all equity options for public customers and firm proprietary traders traded at the Chicago Board Options Exchange (CBOE).<sup>5</sup> The sample covers 29,037 different option series on 1,620 underlyings, with the number of open buy, close buy, open sell, and close sell positions summarized for July 7, 2006. To calculate the net demand on this day per underlying, we calculate the net amount of options bought (open buys plus close buys) and subtract the net amount sold (open sells plus close sells) for every underlying. This measure is the amount of options sold (if positive) or bought (if negative) by the market makers for one underlying on the observation date.

## [Insert Figure 1 about here]

Figure 1 shows the distribution of these net demand measures for different stocks. The median is -2 (mean 19) and is close to zero compared to the huge standard deviation (862) of the net demand. Although our sample is small compared to the sample of Ni, Pan, and Poteshman (2008) and Muravyev (2016), our results show the same pattern of average net demand being close to zero, with a very large standard deviation.<sup>6</sup>

As an alternative to the double sorting, we now look at the relation between stock illiquidity and option expensiveness in a regression framework. Every month, we run a cross-sectional regression of HV–IV on different variants of the Amihud measure. Table II shows the time-series averages of the regression coefficients, with their t-statistics in parentheses. If we just use the Amihud measure, as in model (1), the coefficient is negative, which is in line with the decreasing mean values in the second to last column of Table I. In contrast, the Amihud measure multiplied by the sign function of HV–IV, as in model (2), has a positive and highly statistically significant

<sup>&</sup>lt;sup>4</sup> Since the absolute measurement error for volatility increases with the volatility level and average stock volatility increases with illiquidity, some of the increased dispersion of HV–IV could be caused by increasing measurement error divergence. Such an effect could be alleviated by using the logarithms of the historical and implied volatility. However, our results are qualitatively unchanged when using log(HV/IV).

<sup>&</sup>lt;sup>5</sup> The sample was available on the website of Market Data Express, LLC (November 1, 2014), at <u>http://www.marketdataexpress.com/User Data/Files/openclose\_20060707.zip</u>.

<sup>&</sup>lt;sup>6</sup> The results of Carr and Wu (2009) on variance risk premiums provide complementary evidence, because premiums on individual stocks show large cross-sectional variation.

impact on HV–IV. If we include the Amihud measure multiplied by the sign function in addition to the raw Amihud measure, as in model (3), the latter becomes insignificant. Finally, model (4) uses the Amihud measure multiplied by a dummy variable that is equal to one if HV–IV is positive and by a dummy variable that is equal to one if it is negative. The model confirms our observation from Table I that increasing illiquidity leads to higher HV–IV on the condition that HV–IV is positive and to a lower HV–IV on the condition that HV–IV is negative.

## [Insert Table II about here]

In summary, the regression results also show a strong relation between stock illiquidity and option expensiveness. However, it is crucial to distinguish between cheap options and expensive options. For cheap options, higher stock illiquidity is associated with options being even cheaper. For expensive options, higher illiquidity is associated with options being more expensive.

## B. Stock Illiquidity and Option Returns

We next look at the relation between stock illiquidity and option returns. If the observed pattern of stock illiquidity and option expensiveness is indeed due to the higher hedging costs for illiquid underlyings, we should see a similar pattern for stock illiquidity and option returns. However, historical volatility could just be a noisier estimator of the desired future volatility for less liquid stocks, resulting in higher mean absolute estimation errors. If these higher estimation errors were not reflected in the option's market price, that is, the market's volatility forecast does not contain this noise, the pattern could disappear if we move from option expensiveness to option returns.

The results for stock illiquidity and option returns in Table III are based on the same sorting procedure as used for Table I. They show the time-series averages of the mean (equally weighted) monthly option portfolio returns in the different Amihud and HV–IV quintiles and their corresponding t-statistics. The straddle returns in Panel A and the delta-hedged call returns in Panel B are calculated as described in Section I.B.

For the straddles, we find increasing option returns for the highest HV–IV quintile column and decreasing option returns for the column with the lowest HV–IV values. The delta-hedged call returns show a similar structure.<sup>7</sup> Both patterns mirror the increasing and decreasing HV–IV values of Table I for the corresponding columns. The average returns of the 5–1 columns and

<sup>&</sup>lt;sup>7</sup> Delta-hedged put returns also show a similar structure in unreported results.

rows are calculated from a portfolio that is \$1 long in the fifth portfolio and \$1 short in the first portfolio. As shown in the 5–1 column, the returns are monotonically increasing along the Amihud quintiles.

The return difference of being long in the 5–1 strategy in the highest illiquidity quintile and short in the 5–1 strategy in the lowest illiquidity quintile, that is, high Amihud(5–1) minus low Amihud(5–1), is 8.1% for the straddles and 1.4% for the delta-hedged calls. This strategy combines the impact of illiquidity into a single number, which is highly significant for both straddles and delta-hedged calls.<sup>8</sup>

The last two columns in Table III show that the mean option returns only significantly decline along the Amihud quintiles for the delta-hedged calls, while they do not change significantly for the straddle portfolios. The standard deviation smoothly increases for straddles and delta-hedged calls with high significance.

## [Insert Table III about here]

To obtain a deeper understanding of the liquidity effect on option returns, we refine the sorting on our liquidity measure. For Figure 2, we repeat our analysis from Table III but sort the options every month into deciles instead of quintiles on the Amihud measure. The lower plots in Figure 2 show the average delta-hedged call or straddle returns of the Amihud deciles. The overall negative relation between option returns and illiquidity mirrors the findings of Christoffersen et al. (2015) and Karakaya (2014), who reports that returns to selling delta-hedged options increase with higher underlying stock illiquidity. However, this negative relation is almost completely driven by the highest illiquidity decile, while there is no clear pattern along the remaining deciles. In contrast, a clear pattern emerges once we sort option observations within our double-sorted portfolios, we limit our analysis to HV–IV terciles and display the returns on the 3–1 portfolios. The upper plots in Figure 2 reveal a clear positive trend with stock illiquidity. Even for the lowest illiquidity decile, however, the returns of the 3–1 portfolios are still positive, with a return of

<sup>&</sup>lt;sup>8</sup> From a theoretical perspective, other developments of option returns and HV–IV are also plausible. Assume, for example, a situation in which market makers are always net short on all options. In such a market, all options would be, with our theoretical argument, more expensive when the underlying is more costly to trade. We should then find decreasing returns along the Amihud quintiles, not only for the lowest but also for the highest HV–IV quintiles, and the returns of the 5–1 strategy should not be liquidity driven. In contrast, our empirical evidence that large positive and negative HV–IV values can be found in all liquidity quintiles prompts us to focus on the development of the 5-1 strategy along the illiquidity quintiles.

0.55% and a t-statistic of 2.58 for the delta-hedged calls and a straddle return of 3.65% with a tstatistic of 2.08. Such a positive return is unlikely to be explained by hedging costs due to stock illiquidity alone, because we would then expect the return of the 3–1 portfolio to vanish for very liquid underlyings. However, other market frictions and market incompleteness, for example, caused by jumps or stochastic volatility, could still prevent perfect hedging. In summary, Figure 2 illustrates the main contribution of the paper. By looking at the dispersion of HV–IV and the returns of HV–IV long–short trading strategies, we have uncovered a clear connection between stock illiquidity and option returns and are able to capture a large proportion of the mean returns of long–short strategies based on HV–IV.

[Insert Figure 2 about here]

## **III.** Potential Explanations for the Main Results

#### A. Option Returns and Risk Factors

So far, we have established an empirical pattern that relates option prices and option returns to the underlying's illiquidity. We now look at different potential explanations. A first idea is that the returns of options portfolios are exposed to common risk factors besides stock illiquidity. After controlling for these risks, illiquidity effects could no longer exist. We therefore check whether the increasing excess returns of the 5-1 HV–IV strategies with greater illiquidity of the underlyings can be explained by common risk factors. We run a time-series regression of the returns from the 5-1 HV–IV portfolios within the liquidity quintiles on several risk control variables.

Especially due to the imperfections in our delta hedge and the monthly holding period of the straddle portfolio, the returns could be related to known patterns in the cross section of stock returns. We control for this potential explanation by including the three factors of Fama and French (1993) and Carhart's (1997) momentum factor in a time-series regression.<sup>9</sup> In addition, we control for aggregate volatility and correlation risk premiums following Cao and Han (2013). For market volatility risk, we include the excess returns of the Coval and Shumway (2001) zero-beta

<sup>&</sup>lt;sup>9</sup> Goyal and Saretto (2009), Schürhoff and Ziegler (2011), Frazzini and Pedersen (2012), Cao and Han (2013), Buraschi, Trojani, and Vedolin (2014), and Christoffersen et al. (2015) also include these four factors as control variables for option returns.

Standard & Poor's (S&P) 500 straddle. We also include the value-weighted average return of (available) zero-beta straddles on the S&P 500 component stocks minus the risk-free rate. Driessen, Maenhout, and Vilkov (2009) show that the returns of an index straddle can be decomposed into the returns of index component straddles and a correlation risk trading strategy. Thus, inclusion of the index straddle and the average of its component straddles can be interpreted as a control for correlation risk. Schürhoff and Ziegler (2011) use the component straddle factor as a proxy for the common idiosyncratic volatility risk premium in their empirical work. Details on our risk factor calculations can be found in Appendix B.

The regression results are presented in Table IV. They show that the loadings on the Fama– French (1993) and momentum factors are insignificant in most cases. The HML and the momentum factor is significant for the delta-hedged call 5–1 strategy in the low liquidity quintile. The loading on the zero-beta S&P 500 straddle is positive and significant for straddles in the low liquidity quintiles. For comparison, Goyal and Saretto (2009) also report for their HV–IV trading strategy a significant positive coefficient for the zero-beta S&P 500 straddle and insignificant coefficients for the Fama–French (1993) and momentum factors. The coefficient of the zero-beta S&P 500 component straddle, which is not included in the analysis of Goyal and Saretto (2009), is insignificant for our sample. Overall, the alphas of the portfolios are all significant and very close to the average raw returns reported in Table III.

## [Insert Table IV about here]

The regression alphas of the 5-1 HV–IV strategies within the high liquidity quintile are significantly lower than the alphas of the 5-1 strategies within the low liquidity quintile. The differences for the alphas of the high and low liquidity straddle and delta-hedged call portfolios are 8.0% and 1.2%, respectively. We conclude that the higher absolute option returns we find for the portfolios with more illiquid underlyings cannot be explained by common risk factors.

Other authors have attributed the returns of volatility trading strategies to uncertainty risk, informed trading, and behavioral biases. Buraschi, Trojani, and Vedolin (2014) suggest an explanation for the returns of volatility strategies that is based on the role of priced disagreement risk, but the returns from disagreement risk strategies are very small compared to the option returns we find. Similarly, stocks with higher illiquidity are more likely to be stocks with more private information being available. Since Easley, O'Hara, and Srinivas (1998) and Pan and

Poteshman (2006) show evidence of informed trading in the options market too, one could argue that our option returns stem from asymmetric information. Theoretical models with competitive risk-neutral market makers consider asymmetric information to be a determinant of bid-ask spreads (Copeland and Galai, 1983; Glosten and Milgrom, 1985; Easley and O'Hara, 1987). However, in such a setting, private information does not lead to excess returns of market makers unless market makers charge an information risk premium in the sense of Easley, Hvidkjaer, and O'Hara (2002). In addition, Christoffersen et al. (2015) have empirically shown that private information is a strong determinant of option bid-ask spreads but not of average option returns. Given this evidence and the results of Buraschi, Trojani, and Vedolin (2014), we do not control for disagreement risk and private information. Goyal and Saretto (2009) hypothesize that the returns to their HV–IV strategies could be caused by investors becoming excessively optimistic (pessimistic) about the future riskiness of a stock after large positive (negative) returns. Similarly, An et al. (2014) show that realized excess stock returns help to predict changes in implied volatility. Their findings are consistent with investors' speculative demand for options and intermediaries hedging constraints. Therefore, the findings are complementary to our main result, that higher stock illiquidity is associated with wider fluctuations of option prices around reference prices expected in perfect market environments.

## B. Impact of Transaction Costs on Option Prices and Returns

In principle, the relation between stock illiquidity and option prices observed in the data is consistent with a demand-based option pricing theory and demand pressure coming from end users, with varying signs across individual equity options. However, the question remains as to whether stock illiquidity can be a viable explanation for the empirical patterns. This would require that realistic illiquidity costs of market makers be compatible with the observed magnitudes of price and return effects. We investigate this issue by conducting a simulation study.

Our analysis is based on Leland's (1985) option pricing approach with discrete-time replication and transaction costs that provides estimates of the maximum potential price impact of the illiquidity of the underlying.<sup>10</sup> We first briefly describe Leland's approach. We then explain how we simulate option prices using this approach under realistic assumptions for transaction costs,

<sup>&</sup>lt;sup>10</sup> Alternative pricing models are presented by Boyle and Vorst (1992) and Cetin et al. (2006). The latter model also considers market impact costs that depend on the trade size, which would likely lead to even greater effects.

hedging frequency, market maker positions, and underlying dynamics. Finally, we compare the resulting simulated HV–IV values, option prices, and option returns with our empirical findings.

## B.1. Options Replication for Illiquid Underlyings

Leland (1985) uses a Black–Scholes setting with proportional transaction costs for the underlying and derives the following modification of the variance used in the Black–Scholes model:

$$\sigma_m^2 = \sigma^2 \left( 1 - \frac{k}{\sigma} \sqrt{\frac{2}{\pi \delta t}} \, sgn(V_{SS}) \right),$$

where  $k = (S_{bid} - S_{ask})/S_{mid}$  denotes the round-trip transaction costs for trading in the underlying,  $\sigma$  is the Black–Scholes volatility, and  $\delta t$  is the time interval between two hedging revisions. The sign function on the option gamma  $(sgn(V_{SS}))$  leads to higher volatility (price) when the market maker has to hedge a short option position and decreases the volatility (price) when the market maker has a long position. The higher (lower) option prices for short (long) positions can be thought of as compensation for the market maker to cover the additional hedging costs due to transaction costs. Leland shows that this modified variance results in an upper (lower) bound of the option price from a discrete-time replication strategy with proportional transaction costs.

Leland's (1985) approach has the interesting feature that the standard deviation of the hedging profit and loss (P&L) is close to the standard error of a discrete-time Black–Scholes hedging strategy without transaction costs. If the market maker adjusts the volatility and therefore the price of the option with Leland's adjustment and uses Leland's delta for hedging, the resulting P&L distribution is, ceteris paribus, close to the P&L distribution in a frictionless market with the usual Black–Scholes pricing and hedging at the same frequency. Using Leland's adjustment for pricing and hedging accounts for transaction costs but does not change the resulting risks of the hedged option position. This enables us to interpret the effect of transaction costs independently of the effects described by Gârleanu, Pedersen, and Poteshman (2009). While their work concentrates on the price effects of unhedgeable risks, the Leland adjustment can be seen as the incremental price change due to transaction costs.

#### B.2. Simulation Design

In our simulation, we consider a market maker who manages options on several underlyings and accounts for transaction costs by using Leland's (1985) adjustment. We simulate 10,000 underlyings following uncorrelated geometric Brownian motions with a volatility  $\sigma$  of 40% and a stock price drift  $\mu$  of 10%. For every underlying, there is one ATM call option with a strike of 100 and a time to maturity of one month. The risk-free rate *r* is 5%. The market maker is either long or short in the call option on one underlying with a 50% chance. When the market maker is trading the underlying, there are transaction costs k/2 that are proportional to the stock price (relative half-spread). The transaction costs are either 0.1%, 0.2%, 0.3%, 0.4%, or 0.5%, all with equal probability across stocks.<sup>11</sup> Market makers adjust the hedge of their short or long positions with the risk-free asset and stocks every day until maturity and account for the hedging costs by using Leland's (1985) adjustment, considering their long or short position in options.

We concentrate on a trading strategy using delta-hedged calls as defined in Section I.B. The expected return for the delta-hedged call according to Eq. (1) is calculated as<sup>12</sup>

$$E(\Pi_{t,t+\tau}) = \frac{E[\max(S_{t+\tau} - K, 0)] - \Delta_{C,t} S_t e^{\mu\tau} - (C_t - \Delta_{C,t} S_t) e^{r\tau}}{Abs(C_t - \Delta_{C,t} S_t)}.$$
(3)

To obtain our results, we first sort options on their underlying liquidity costs into five groups, each consisting of 2,000 option observations, where we assume that the market maker is long on one half and short on the other. Within these groups, we sort the options again into quintiles based on their HV–IV values, where the historical volatilities are estimates based on one year of simulated daily return data with a true return volatility of 40%.

#### **B.3.** Simulation Results

Table V shows the results for the simulated data, which correspond to the results of Tables I and III. Panel A reports the average HV–IV values and Panel B the average returns. The results are very similar to those obtained for the market data. The average absolute HV–IV values of the low and high HV–IV quintiles increase with higher transaction costs, while the mean HV–IV values of all options along the transaction cost quintiles are comparably small. The results for the returns

<sup>&</sup>lt;sup>11</sup> Bessembinder (2003) reports large, medium, and small New York Stock Exchange stocks' average quoted half bid–ask spreads, which are equal to 0.2%, 0.5%, and 0.8%, respectively.

<sup>&</sup>lt;sup>12</sup> We calculate the expected option payoff under the P-measure based on a geometric Brownian motion for the stock price process with 40% volatility and a drift rate of 10% per year.

of delta-hedged calls show a similar picture. As for our market data, the returns on the long–short (5–1) strategy are much higher in the high transaction cost category than in the low transaction cost category. Remarkably, the magnitudes of the average HV–IV differences and delta-hedged return differences between the quintiles are similar to those observed in the empirical data.

The penultimate column of Table V shows the mean HV–IV and mean returns of all options within one transaction cost group. The differences between these means are relatively small across the transaction cost groups. Thus, in our scenario, when market makers are equally likely net long or net short, the effect of transaction costs cannot be seen from the interaction of underlying transaction costs with average HV–IV expensiveness or option returns alone. The last column of Table V reports the standard deviation of HV–IV and option returns within the transaction costs, as in Section II. Only the magnitude of the standard deviations is smaller, since we do not account for variation in true volatility, which we fixed at 40%, and use fixed expected option returns according to Eq. (3).

## [Insert Table V about here]

For Figure 2, we used extended sorting on the liquidity measure. Every month, option observations were sorted into deciles based on their underlying Amihud measure. Within these deciles, the options were again sorted by HV–IV into terciles. We now follow the same procedure with the simulated data. Figure 3 shows the results by depicting both the empirical average monthly delta-hedged returns of the 3–1 long–short strategy and the corresponding ones from the simulation. We see that the pattern is very similar and the magnitude of the simulated delta-hedged call returns also comes close to the average empirical returns.

## [Insert Figure 3 about here]

We conclude that our empirical results for option expensiveness and option returns under the double sorting with respect to the Amihud measure and HV–IV can be reproduced by a simple simulation with realistic transaction cost assumptions, a market maker being equally likely long or short in options on one underlying and accounting for transaction costs in a simple way with Leland's (1985) adjustment.

## **IV. Robustness Checks**

The previous analysis raises different questions about the robustness of our main results. A first issue is the measurement of option expensiveness, particularly the estimation of historical volatility, and the measurement of illiquidity. We deal with this problem in Section IV.A. A second robustness issue refers to the time periods considered. This issue has two aspects: the chosen return period and the chosen sample period. Sections IV.B and IV.C, respectively, deal with these points. A third robustness issue refers to the underlying rationale for our results. If it is really demand pressure and the fact that market makers are long in some and short in other specific option series that drives our findings, we should not only find effects related to general expensiveness, as measured by HV-IV, but also similar illiquidity effects for the relative expensiveness of puts and calls, given that market makers could be on different sides of the market in both types of options. This issue is investigated in Section IV.D.

## A. Alternative Volatility and Illiquidity Measures

The analysis in Section 3 can be criticized due to measurement problems. First, the specific historical volatility used could be an inadequate benchmark for measuring option expensiveness. Second, because illiquidity is a multidimensional phenomenon, our results could depend on the specific illiquidity measure used. We investigate these issues in different robustness analyses.

As alternative volatility measures, we use a GARCH(1,1) estimate for the option's lifetime volatility and the standard deviation of daily stock returns using the six and 24 most recent months.<sup>13</sup> The results are presented in Table VI. There is no clear pattern in the mean expensiveness resulting from these alternative measures: the change of the average HV–IV with illiquidity is significantly positive for the GARCH(1,1) estimate, whereas it is significantly negative for the six-month estimate and insignificant for the 24-month estimate. In contrast, the increase of the standard deviation is highly significant for all volatility measures. Overall, these findings are in line with the results and conclusions from Table I.

## [Insert Table VI about here]

Another doubt about the previous results is that they could be tied to our illiquidity measure. For Table VII, we repeat our analysis of Table I but replace the Amihud illiquidity measure with

<sup>&</sup>lt;sup>13</sup> Details on the GARCH(1,1) estimation process can be found in Appendix C.

alternative measures: the log market capitalization of the underlying stock, the dollar trading volume of the underlying, and Roll's (1984) and Corwin and Schultz's (2012) bid–ask spread estimates. The picture of an increasing standard deviation of HV–IV with increasing illiquidity remains unchanged for all alternative measures, while the mean HV–IV is either positively or negatively correlated with illiquidity, depending on the measure.

## [Insert Table VII about here]

In Table VIII, we check if the increasing absolute HV–IV values with the alternative historical volatility and illiquidity measures translate into higher excess returns. We repeat the analysis from Table IV, but with the alternative measures. The differences of the alphas from the 5–1 strategy in the lowest liquidity quintile compared to the 5–1 strategy in the highest liquidity quintile are significant with all alternative liquidity and volatility measures.

## [Insert Table VIII about here]

#### B. Daily Returns

So far, we have used monthly option returns, as described in Section I.B. During the one-month holding period, option moneyness could change drastically and the returns of the delta-hedged calls could be exposed to substantial underlying stock price risk. We therefore repeat our analysis from Table III with daily delta-hedged call and straddle returns. The delta-hedged call return calculation is similar to that presented in Eq. (1). Instead of holding the option until maturity and calculating the option payoff, we use the option mid price of the day following the trading initiation date t. Similarly, we modify the straddle return calculation from Eq. (2) by replacing the option payoffs with the next day's option mid prices and adjust the funding costs for a one-day holding period.

The resulting option returns in Table IX are qualitatively the same as in Table III. Again, the returns in the low and high HV–IV columns smoothly decrease and increase, respectively, across the Amihud quintiles. In addition, the increasing profitability of the 5–1 HV–IV strategy with higher stock illiquidity is highly significant, especially for the delta-hedged call returns; the significance increases and now has a t-statistic above 10. Interestingly, the option return of the first day seems to capture a large fraction of the monthly option returns reported in Table III.

[Insert Table IX about here]

#### C. Alternative Sample Periods

Until 1999, options were often listed only on one exchange, which governed all interactions between market participants. In October 1999, the U.S. Securities and Exchange Commission (SEC) ordered the option exchanges to develop a plan to electronically link the various market centers. Battalio, Hatch, and Jennings (2004) have shown that option market efficiency improved during this period, in which the equity option market evolved toward a national market system. The final implementation of the SEC's options exchange linkage plan and more stringent quoting and disclosure rules became effective in April 2003. We therefore check if our results are driven by market inefficiencies before these structural changes took place and exclude the period before May 2003 from our analysis. In a next step, we also exclude the period during the financial crisis to ensure that the market turmoil in this period does not drive our results.

The portfolio construction and return calculation for Table X are the same as for Table III. The first column returns correspond to the 5–1 column returns in Table III. The second and third columns exclude observations before the option market structure changes up to May 2003 and the third column additionally excludes the financial crises from June 2007 to December 2009.

The difference of the portfolio returns between the highest and lowest illiquidity quantiles for the period May 2003 to August 2015 is very similar to the difference for the complete sample period. Interestingly, the overall performance of the HV-IV trading strategy decreases in all illiquidity quantiles if we exclude the period before the market reforms. The market seems to have become more efficient, while the link between stock illiquidity and option returns has remained stable.

## [Insert Table X about here]

#### D. Relative Expensiveness of Puts and Calls

So far, we have looked at average expensiveness and have not considered possible distortions in the relative pricing of calls and puts due to stock illiquidity. However, stock illiquidity could well drive a wedge between put and call prices. A natural measure for this discrepancy is the difference between put and call implied volatility. If calls are cheap compared to corresponding puts, this could reflect a situation with a relatively low or, let us say, negative net demand for calls and a relatively high demand for puts. In such a situation, the difference between put and call implied volatilities (PVOL–CVOL) is positive and higher stock illiquidity is expected to decrease call prices even further relative to their corresponding put prices. Conversely, if call

prices are relatively expensive, the negative PVOL–CVOL measure should decrease even further with higher stock illiquidity. We therefore repeat our analysis from Table I but use the difference between put and call implied volatilities (PVOL–CVOL) as a measure of (relative) expensiveness. For the new analysis, we exclude all put and call combinations for one underlying where the moneyness does not exactly match. The sample size is therefore slightly smaller, with 156,125 observations, compared to the original straddle sample with 156,849 observations.

## [Insert Table XI about here]

Table XI shows our results. The absolute deviations between the two implied volatility measures indeed increase with stock illiquidity. Consistent with the above arguments, we also find higher negative and positive differences within the higher illiquidity quantile. The column with the mean PVOL–CVOL values of all observations within the Amihud quantiles shows a significant tendency toward more expensive puts with higher illiquidity.

To analyze the return implications, we next consider a portfolio consisting of an ATM call long and an ATM put short (synthetic future) delta hedged with a short position in the stock. The return of this portfolio is

$$\Pi_{t,t+\tau} = \frac{\max(S_{t+\tau} - K_c, 0) - \max(K_P - S_{t+\tau}, 0) - S_{t+\tau} - (C_t - P_t - S_t)e^{r\tau}}{Abs(C_t - P_t - S_t)}$$

Since our strikes are equal  $(K_C = K_P)$ , this simplifies to

$$\Pi_{t,t+\tau} = \frac{(S_t - C_t + P_t)e^{r\tau} - K}{Abs(C_t - P_t - S_t)}$$

The returns then do not depend on the stock's return until maturity. Table XII shows the resulting returns of an equal-weighted investment in the illiquidity and PVOL–CVOL categories. The structure of the returns is as we would expect from the structure in Table XI. The differences between the highest and lowest illiquidity quantiles are highly significant, showing that stock illiquidity has an impact on the relative pricing of puts and calls, which is in accordance with the idea of order imbalances in put and call options.

#### [Insert Table XII about here]

## V. Conclusions

This paper is the first to present empirical evidence that underlying stock illiquidity is strongly related to option expensiveness and option returns. We show in a cross-sectional analysis that the standard deviation of our option expensiveness measure HV–IV increases with stock illiquidity, while the mean HV–IV does not change with illiquidity. The documented pattern also translates into economically and statistically significant delta-hedged call and straddle excess returns, showing that stock illiquidity can capture a large part of the option returns determined by Goyal and Saretto (2009). The results are qualitatively unchanged for different measures of option expensiveness and stock illiquidity. Moreover, similar illiquidity effects hold for the relative expensiveness of put and call options.

Our findings are in line with intermediaries considering different option hedging costs depending on stock liquidity and being net long in options on some stocks and short in options on others, a setting supported by an empirical sample of end user net demand data. A simulation study shows that if an intermediary is equally likely to be long or short in options on one underlying and accounts for realistic hedging costs when setting options prices, the resulting deviations of historical and implied volatility as well as the resulting option returns are strikingly similar to those observed in our empirical data.

Our results suggest that stock illiquidity plays an important role in the explanation of the difference between historical and implied volatility. Theoretically, the effect depends on the sign of the end users' net demand, which intermediaries have to supply. Therefore, our analysis provides a rationale for using the sign of HV–IV as a proxy for the sign of end users' net demand, which offers new ways to study the impact of demand on option prices.

#### **Table I**

#### Historical minus implied volatility (HV-IV) of two-way sorted portfolios.

The sample between January 1996 and August 2015 includes 156,849 pairs of call and put options. Each month option observations are first sorted into quintiles based on the Amihud illiquidity measure. Within these quintiles, options are sorted into quintiles based on the difference between the historical and implied volatility. This table shows the average difference between historical volatility and implied volatility for the different categories. The historical volatility is the standard deviation of daily stock returns using the 12 most recent months. The implied volatility is the average of the call and put implied volatilities. Associated t-statistics are corrected for autocorrelation following Newey and West (1987).

Average HV–IV										
			mean	sd						
	1-low	2	3	4	5-high	5-1	t-stat.		a	11
1-low	-7.0%	-1.5%	1.2%	4.2%	12.4%	19.4%	17.52		1.8%	7.7%
2	-9.0%	-2.3%	0.8%	4.1%	13.4%	22.5%	18.66		1.4%	9.0%
3	-10.9%	-2.9%	0.9%	4.8%	15.5%	26.3%	22.13		1.5%	10.6%
4	-12.8%	-3.3%	0.9%	5.5%	17.5%	30.2%	25.69		1.6%	12.4%
5-high	-16.2%	-4.4%	0.8%	6.2%	20.3%	36.4%	30.44		1.3%	15.0%
5–1 t-stat.	-9.2% -17.55	-2.9% -7.30	-0.4% -1.18	2.0% 5.31	7.9% 12.06	17.1% 23.88			-0.5% -1.27	7.3% 14.48
	1-low 2 3 4 5-high 5–1 t-stat.	1-low         1-low           2         -9.0%           3         -10.9%           4         -12.8%           5-high         -16.2%           5-1         -9.2%           t-stat.         -17.55	$\begin{tabular}{ c c c c c c } \hline Ave \\ \hline Historical \\ \hline 1-low & 2 \\ \hline 1-low & -7.0\% & -1.5\% \\ 2 & -9.0\% & -2.3\% \\ 3 & -10.9\% & -2.9\% \\ 4 & -12.8\% & -3.3\% \\ 5-high & -16.2\% & -4.4\% \\ \hline 5-1 & -9.2\% & -2.9\% \\ t-stat. & -17.55 & -7.30 \\ \hline \end{tabular}$	Average HV-Historical - implied1-low231-low-7.0%-1.5%1.2%2-9.0%-2.3%0.8%3-10.9%-2.9%0.9%4-12.8%-3.3%0.9%5-high-16.2%-4.4%0.8%5-1-9.2%-2.9%-0.4%t-stat17.55-7.30-1.18	Average HV–IVHistorical - implied volatility1-low2341-low-7.0%-1.5%1.2%4.2%2-9.0%-2.3%0.8%4.1%3-10.9%-2.9%0.9%4.8%4-12.8%-3.3%0.9%5.5%5-high-16.2%-4.4%0.8%6.2%5-1-9.2%-2.9%-0.4%2.0%t-stat17.55-7.30-1.185.31	Average HV–IVHistorical - implied volatility (HV–IV)1-low2345-high1-low-7.0%-1.5%1.2%4.2%12.4%2-9.0%-2.3%0.8%4.1%13.4%3-10.9%-2.9%0.9%4.8%15.5%4-12.8%-3.3%0.9%5.5%17.5%5-high-16.2%-4.4%0.8%6.2%20.3%5-1-9.2%-2.9%-0.4%2.0%7.9%t-stat17.55-7.30-1.185.3112.06	Average HV-IVHistorical - implied volatility (HV-IV)1-low2345-high5-11-low-7.0%-1.5%1.2%4.2%12.4%19.4%2-9.0%-2.3%0.8%4.1%13.4%22.5%3-10.9%-2.9%0.9%4.8%15.5%26.3%4-12.8%-3.3%0.9%5.5%17.5%30.2%5-high-16.2%-4.4%0.8%6.2%20.3%36.4%5-1-9.2%-2.9%-0.4%2.0%7.9%17.1%t-stat17.55-7.30-1.185.3112.0623.88	Average HV–IVHistorical - implied volatility (HV–IV)1-low2345-high5–1t-stat.1-low-7.0%-1.5%1.2%4.2%12.4%19.4%17.522-9.0%-2.3%0.8%4.1%13.4%22.5%18.663-10.9%-2.9%0.9%4.8%15.5%26.3%22.134-12.8%-3.3%0.9%5.5%17.5%30.2%25.695-high-16.2%-4.4%0.8%6.2%20.3%36.4%30.445–1-9.2%-2.9%-0.4%2.0%7.9%17.1%t-stat17.55-7.30-1.185.3112.0623.88	Average HV–IVHistorical - implied volatility (HV–IV)1-low2345-high5–1t-stat.1-low-7.0%-1.5%1.2%4.2%12.4%19.4%17.522-9.0%-2.3%0.8%4.1%13.4%22.5%18.663-10.9%-2.9%0.9%4.8%15.5%26.3%22.134-12.8%-3.3%0.9%5.5%17.5%30.2%25.695-high-16.2%-4.4%0.8%6.2%20.3%36.4%30.445–1-9.2%-2.9%-0.4%2.0%7.9%17.1%t-stat17.55-7.30-1.185.3112.0623.88	Average HV–IVmeanHistorical - implied volatility (HV–IV)mean1-low2345-high5–1t-stat.a1-low-7.0%-1.5%1.2%4.2%12.4%19.4%17.521.8%2-9.0%-2.3%0.8%4.1%13.4%22.5%18.661.4%3-10.9%-2.9%0.9%4.8%15.5%26.3%22.131.5%4-12.8%-3.3%0.9%5.5%17.5%30.2%25.691.6%5-high-16.2%-4.4%0.8%6.2%20.3%36.4%30.441.3%5-1-9.2%-2.9%-0.4%2.0%7.9%17.1%-0.5%t-stat17.55-7.30-1.185.3112.0623.88-1.27

#### Table II

#### Regressions of HV-IV on different variations of the Amihud illiquidity measure.

The sample between January 1996 and August 2015 includes 156,849 pairs of call and put options. This table reports the average coefficients from monthly Fama-MacBeth regressions of HV–IV on a constant and the Amihud illiquidity measure multiplied by the sign function of HV–IV, a dummy if HV–IV is positive or negative, or the identity function 1. HV-IV is calculated for the pairs of call and put options as in Table 1. Newey and West (1987) t-statistics are given in parentheses.

			Amil	nud x	
	const.	sign (HV–IV)	(HV-IV) > 0	(HV-IV) < 0	1
(1)	0.016				-0.180
	(1.64)				(-2.30)
(2)	0.016	2.551			
	(1.65)	(12.61)			
(3)	0.015	5.153			-0.560
	(1.48)	(7.22)			(-0.67)
(4)	0.015		4.593	-5.713	
< /	(1.48)		(11.43)	(-3.82)	

#### **Table III**

Average monthly post-formation returns of two-way sorted portfolios.

The sample between January 1996 and August 2015 includes 156,849 pairs of call and put options for the straddle returns. For the delta-hedged call returns, the sample includes 197,374 calls. Each month, option observations are first sorted into quintiles based on the Amihud liquidity measure. Within these quintiles, options are sorted into quintiles based on the difference between the historical and implied volatility. This table shows the average monthly returns of the portfolios for the different categories. The portfolio returns use an equal weighting of the returns of all straddles (delta-hedged calls) falling in the category. For the return calculation, the average of the closing bid and ask quotes is the reference beginning price. The terminal payoff of the options depends on the stock price and the strike price of the option. The hedge ratio for the delta-hedged calls is determined from the implied volatility at trading initiation. Associated t-statistics are corrected for autocorrelation following Newey and West (1987).

#### Panel A

		mean	sd							
	1-low 2 3 4 5-high 5–1 t-stat.									11
	1-low	-4.5%	-1.3%	0.6%	-0.3%	2.9%	7.4%	3.66	-0.5%	74.9%
	2 2	-4.5%	-2.9%	1.4%	-0.7%	4.4%	8.9%	4.99	-0.4%	75.6%
	3	-6.1%	-0.3%	0.2%	1.9%	4.5%	10.5%	5.49	0.0%	77.0%
, <b>,</b>	र ४	-6.7%	-1.0%	0.6%	3.4%	7.7%	14.4%	6.50	0.8%	80.7%
	5-high	-9.7%	-3.4%	-3.8%	-0.2%	5.8%	15.5%	8.07	-2.2%	81.6%
	5-1	-5.1%	-2.1%	-4.4%	0.1%	3.0%	8.1%		-1.7%	6.7%
	t-stat.	-3.18	-1.09	-2.16	0.06	1.56	3.68		-1.40	4.91

Panel B

			Delta-he	dged call	returns					
				mean	sd					
	1-low 2 3 4 5-high 5-1 t-stat.								a	11
	1-low	-0.4%	-0.1%	0.0%	0.0%	0.4%	0.7%	2.97	0.0%	6.3%
pn	2	-0.4%	-0.2%	0.0%	0.1%	0.5%	1.0%	4.54	0.0%	7.1%
nih	3	-0.7%	-0.1%	0.0%	0.4%	0.6%	1.3%	5.34	0.0%	8.3%
Ar	4	-1.1%	-0.2%	0.1%	0.3%	0.9%	2.0%	6.19	0.0%	9.2%
	5-high	-1.5%	-0.6%	-0.4%	-0.2%	0.6%	2.1%	8.48	-0.4%	10.6
	5-1	-1.1%	-0.5%	-0.3%	-0.2%	0.3%	1.4%		-0.4%	4.3%
	t-stat.	-4.28	-2.46	-2.05	-1.14	1.05	4.75		-2.51	21.7

#### **Table IV**

#### Risk-adjusted straddle and delta-hedged call returns.

This table presents the coefficients and t-statistics of a time-series regression of the portfolio returns on the Fama and French (1993) factors (MKT-Rf, SMB, HML), the Carhart (1997) momentum factor (MOM), the Coval and Shumway (2001) excess zero-beta S&P 500 straddle factor (ZB-STR–Index), and the value-weighted average of the zero-beta straddles of the S&P 500 components (ZB-STR–Stocks). The 5-1 portfolios from the highest and lowest liquidity quintiles are constructed as in Table III. The t-statistics for the coefficients in brackets are calculated with Newey and West (1987) standard errors.

		Straddle	S	Delta-hedged calls				
	5–1	5-1	5–1 low liq	5-1	5-1	5–1 low liq		
	high liq.	low lıq.	5–1 high liq.	hıgh lıq.	low liq.	5–1 high liq.		
Alpha	0.077	0.159	0.081	0.009	0.021	0.012		
	(3.89)	(8.47)	(3.68)	(3.33)	(8.78)	(4.29)		
MKT–Rf	-0.495	0.035	0.530	-0.086	0.059	0.145		
	(-0.56)	(0.07)	(0.74)	(-0.89)	(0.82)	(1.85)		
SMB	-0.373	1.032	1.404	-0.132	-0.048	0.083		
	(-0.35)	(1.61)	(1.44)	(-0.86)	(-0.50)	(0.47)		
HML	1.054	0.743	-0.311	0.029	0.206	0.177		
	(1.48)	(1.34)	(-0.48)	(0.24)	(2.02)	(1.23)		
MOM	0.794	0.805	0.010	0.009	0.200	0.192		
	(1.32)	(1.61)	(0.02)	(0.09)	(3.22)	(1.84)		
ZB-STR-	0.008	0.082	0.074	0.005	0.008	0.002		
Index	(0.14)	(2.17)	(1.21)	(0.93)	(1.58)	(0.33)		
ZB-STR-	0.092	0.027	-0.065	-0.005	-0.004	0.001		
Stocks	(0.67)	(0.31)	(-0.43)	(-0.38)	(-0.35)	(0.08)		

#### Table V

Average HV–IV and average expected delta-hedged call returns of two-way sorted portfolios, historical volatility estimated from simulated data, and implied volatility from Leland's adjustment.

For the simulation, we use 1,000 options for every combination of transaction costs (k/2 = 0.1%, 0.2%, 0.3%, 0.4%, or 0.5%) and market maker position (long or short). Within the transaction cost groups k, every option is assigned to a quintile based on the difference between historical and implied volatility (HV–IV). The historical volatility is measured from one year of simulated daily returns (with  $\sigma = 40\%$ ) for every option. The implied volatility is the one resulting from Leland's adjustment. Panel A shows the average HV–IV values for all combinations of transaction costs and HV–IV quintiles. Panel B shows the average expected delta-hedged returns of the options for all combinations of transaction costs and HV–IV quintiles.

Panel A

			Average	e HV-IV					
			mean	sd					
		5-1	al	11					
u	1-low	-3.1%	-1.2%	0.0%	1.3%	3.0%	6.1%	0.0%	2.2%
ctio s	2	-4.4%	-2.1%	0.0%	1.9%	4.1%	8.5%	-0.1%	3.1%
nsad	3	-5.8%	-3.5%	-0.3%	3.1%	5.3%	11.1%	-0.2%	4.2%
[rai	4	-7.3%	-4.9%	-0.3%	4.4%	6.6%	13.9%	-0.3%	5.5%
	5-high	-8.7%	-6.5%	-0.6%	5.5%	7.7%	16.4%	-0.5%	6.7%
	5-1	-5.6%	-5.3%	-0.6%	4.3%	4.6%	10.3%	-0.5%	4.5%
	5-high 5-1	-8.7% -5.6%	-6.5% -5.3%	-0.6% -0.6%	5.5% 4.3%	7.7% 4.6%	16.4% 10.3%	-0.5% -0.5%	6.7% 4.5%

Panel B

Average delta-hedged call returns

		8	Historica	l - implied	l volatility	(HV-IV)		mean	sd
		1-low	5-1	a	1				
ц	1-low	-0.2%	-0.1%	0.0%	0.1%	0.2%	0.5%	0.0%	0.3%
s	2	-0.6%	-0.5%	0.0%	0.6%	0.6%	1.2%	0.0%	0.6%
iost	3	-0.9%	-0.9%	0.1%	0.9%	0.9%	1.8%	0.0%	0.9%
lrar c	4	-1.1%	-1.1%	0.1%	1.3%	1.3%	2.4%	0.1%	1.2%
	5-high	-1.4%	-1.4%	0.1%	1.6%	1.6%	3.0%	0.1%	1.5%
	5-1	-1.2%	-1.3%	0.1%	1.5%	1.4%	2.5%	0.1%	1.2%

#### **Table VI**

#### Average HV-IV with alternative volatility measures.

The sample between January 1996 and August 2015 includes 156,849 pairs of call and put options. Each month, option observations are sorted into quintiles based on the Amihud measure. The table shows the time-series average mean and standard deviation of the difference between historical volatility and implied volatility for the different illiquidity quintiles. The implied volatility is the average of the call and put implied volatilities. The historical volatility in HV–IV is defined with alternative volatility estimates. The first two columns utilize a GARCH(1,1) model using at least one month and a maximum of five years of daily return data. The remaining columns use the standard deviation of the most recent six months and two years of daily returns. Associated t-statistics are corrected for autocorrelation following Newey and West (1987).

		GARC	H (1,1)	6-month		6-month		2-у	vear
		mean	sd	mean	sd	mean	sd		
	1-low	-1.3%	10.3%	1.1%	7.6%	2.9%	8.4%		
pn	2	-0.9%	13.0%	0.7%	9.0%	2.5%	9.7%		
nih	3	0.1%	15.9%	0.6%	10.7%	2.7%	11.3%		
Aı	4	1.3%	18.0%	0.6%	12.4%	3.0%	13.1%		
	5-high	2.5%	21.2%	-0.2%	15.1%	3.0%	15.9%		
	5–1 t-stat.	3.7% 6.73	10.9% 16.10	-1.3% -3.31	7.5% 16.92	0.1% 0.30	7.4% 17.80		

#### Table VII

#### Average HV–IV with alternative stock illiquidity measures.

The sample between January 1996 and August 2015 includes 156,849 pairs of call and put options. Each month, option observations are sorted into quintiles based on different illiquidity measures. The table shows the time-series mean and standard deviation of the difference between historical volatility and implied volatility for the different illiquidity quintiles. The implied volatility is the average of the call and put implied volatilities. The historical volatility in HV–IV is the standard deviation of the most recent 12 months of daily returns. The logarithm of the stock's market capitalization [ln(Size)] and the stock's dollar trading volume [Dollar Volume] are measured on the portfolio formation date. The Roll (1984) [Roll] and Corwin and Schultz (2012) [Corwin-Schultz] measures are estimated from daily stock price data of the month preceding the portfolio formation date. Associated t-statistics are corrected for autocorrelation following Newey and West (1987).

	ln(Size)	Dollar Volume	Roll	Corwin-Schultz		
	mean sd	mean sd	mean sd	mean sd		
1-low	1.9% 16.1%	1.2% 13.7%	1.0% 8.7%	0.7% 7.5%		
2	1.6% 11.9%	1.5% 11.8%	1.1% 8.6%	0.6% 7.8%		
3	1.4% 10.6%	1.5% 11.1%	1.4% 9.5%	1.1% 9.5%		
4	1.3% 8.5%	1.6% 9.9%	2.0% 10.9%	2.1% 11.9%		
5-high	1.5% 7.1%	2.0% 9.2%	3.4% 14.9%	3.3% 16.1%		
5–1	-0.5% -9.0%	0.8% -4.5%	2.4% 6.2%	2.6% 8.7%		
t-stat.	-1.16 -25.39	2.08 -8.60	4.19 13.08	4.08 21.94		

#### **Table VIII**

# Risk-adjusted straddle and delta-hedged call returns with alternative volatility measures and stock illiquidity definitions.

This table presents the alphas (t-statistics) of a time-series regression of the portfolio returns on the Fama and French (1993) factors, the Carhart (1997) momentum factor, the Coval and Shumway (2001) excess zero-beta S&P 500 straddle factor, and the value-weighted average of the zero-beta straddles of the S&P 500 components. The 5–1 portfolios from the highest and lowest liquidity quintiles are constructed as in Table IV but in Panel A the HV measure is replaced with alternative volatility estimates and in Panel B alternative illiquidity measures are used instead of the Amihud measure. Panel B uses the same HV measure as in Table IV. The t-statistics for the coefficients in brackets are calculated with Newey and West (1987) standard errors.

		Straddle	S	Delta-hedged calls				
	5–1	5–1	5–1 low liq	5—1	5–1	5–1 low liq		
	high liq.	low liq.	5–1 high liq.	high liq.	low liq.	5–1 high liq.		
Panel A: Alternative	volatility estim	ates						
GARCH(1,1)	0.018	0.097	0.079	0.004	0.012	0.008		
	(0.88)	(4.65)	(3.07)	(1.54)	(4.46)	(2.33)		
6-month	0.071	0.144	0.073	0.007	0.020	0.012		
	(3.16)	(7.27)	(2.71)	(2.93)	(8.19)	(3.79)		
2-year	0.070	0.127	0.056	0.007	0.017	0.010		
	(3.74)	(6.64)	(2.42)	(3.18)	(7.12)	(3.75)		
Panel B: Alternative i	lliquidity meas	sures						
ln(Size)	big	small	small–big	big	small	small–big		
	0.103	0.171	0.068	0.011	0.022	0.011		
	(5.81)	(8.81)	(3.02)	(5.32)	(7.66)	(3.47)		
Dollar Volume	high	low	low-high	high	low	low-high		
	0.078	0.160	0.082	0.011	0.021	0.010		
	(4.13)	(8.78)	(3.73)	(3.66)	(9.19)	(3.12)		
Roll	low	high	high–low	low	high	high–low		
	0.095	0.166	0.071	0.009	0.026	0.017		
	(4.73)	(7.70)	(2.65)	(3.95)	(6.49)	(4.23)		
Corwin–Schultz	low	high	high–low	low	high	high–low		
	0.116	0.165	0.049	0.010	0.028	0.018		
	(4.98)	(10.32)	(1.83)	(4.87)	(8.07)	(5.13)		

#### **Table IX**

Average daily post-formation returns of two-way sorted portfolios.

The sample between January 1996 and August 2015 includes 156,849 pairs of call and put options for the straddle returns. For the delta-hedged call returns, the sample includes 197,374 calls. Each month, option observations are first sorted into quintiles based on the Amihud liquidity measure. Within these quintiles, options are sorted into quintiles based on the difference between the historical and implied volatility. This table shows the average daily returns of the portfolio for the different categories. The portfolio returns use an equal weighting of the returns of all straddles (delta-hedged calls) falling in the category. For the return calculation, the average of the closing bid and ask quotes is the reference beginning price. The option positions are closed at the average of the closing bid and ask quotes on the following trading day. The hedge ratio for the delta-hedged calls is determined from the implied volatility upon trading initiation. Associated t-statistics are corrected for autocorrelation following Newey and West (1987).

			Stra	ddle retu	rns					
Historical - implied volatility (HV–IV)										sd
1-low 2 3 4 5-high 5-1 t-stat.									6	ıll
	1-low	-1.26%	-0.72%	-0.55%	-0.08%	-0.23%	1.03%	5.28	-0.6%	7.3%
pn	2	-1.45%	-0.68%	-0.59%	-0.10%	-0.14%	1.31%	7.57	-0.6%	7.1%
mih	3	-1.49%	-0.63%	-0.35%	0.08%	-0.15%	1.34%	8.25	-0.5%	7.5%
Ā	4	-1.58%	-0.54%	-0.20%	0.04%	0.04%	1.62%	9.63	-0.4%	7.6%
	5-high	-1.99%	-0.82%	-0.32%	0.00%	0.26%	2.25%	12.01	-0.6%	8.2%
	5-1	-0.73%	-0.10%	0.22%	0.08%	0.49%	1.22%		0.0%	1.0%
	t-stat.	-3.64	-0.61	1.09	0.40	2.28	4.65		-0.07	4.89

Delta-hedged call returns Historical - implied volatility (HV-IV) mean sd 2 1-low 3 4 5-high 5-1 all t-stat. 1-low 0.04% -0.05% 0.01% 0.02% 0.02% -0.02% -0.28 0.0% 1.1% Amihud 2 -0.19% -0.02% -0.01% 0.01% 0.06% 0.24% 8.16 0.0% 0.9% 3 0.07%0.09% -0.23% -0.05% -0.02% 0.32% 9.27 0.0% 1.1% 4 -0.33% -0.06% -0.03% 0.05% 0.11% 0.44% -0.1% 13.56 1.1% -0.72% -0.17% -0.01% 0.29% -0.1% 5-high 0.14% 1.01% 16.35 1.6% 5 - 1-0.76% -0.12% -0.02% 0.11% 0.28% 1.04% -0.1% 0.6% t-stat. -9.92 -3.93 -0.73 3.95 8.55 12.25 -4.46 4.47

#### Table X

#### Post-formation option returns for alternative sample periods.

The sample between January 1996 and August 2015 includes 156,849 pairs of call and put options for the straddle returns. For the delta-hedged call returns, the sample includes 197,374 calls. The sample between May 2003 and August 2015 excludes the period before the SEC's options exchange linkage plan became effective. In addition, the last column excludes the financial crisis between June 2007 and December 2009. Each month, option observations are first sorted into quintiles based on the Amihud liquidity measure. Within these quintiles, options are sorted into quintiles based on the difference between the historical and implied volatility. This table shows the averages of the monthly differences between the returns of the high HV-IV portfolios and the low HV-IV portfolios (5-1) within the Amihud quantiles. The portfolio returns use an equal weighting of the returns of all straddles (Panel A) and deltahedged calls (Panel B) falling in the category. For the return calculation, the average of the closing bid and ask quotes is the reference beginning price. The terminal payoff of the options depends on the stock price and the strike price of the option. Associated t-statistics are corrected for autocorrelation following Newey and West (1987).

		Jan. 1996 -	May 2003 -	May 2003 - Aug. 2015
		Aug. 2015	Aug. 2015	excl. Jun. 2007 - Dec. 2009
Panel A: Straddle return	ns (5-1)			
	1-low	7.4%	3.4%	3.8%
pn	2	8.9%	5.4%	4.4%
hin	3	10.5%	5.6%	5.2%
Aı	4	14.4%	9.4%	7.5%
	5-high	15.5%	11.8%	11.2%
	5–1	8.1%	8.4%	7.4%
	t-stat.	3.68	3.13	2.38
Panel B: Delta-hedged	call returns (5-1)			
	1-low	0.7%	0.2%	0.9%
pn	2	1.0%	0.5%	1.1%
nih	3	1.3%	0.5%	1.5%
Ar	4	2.0%	1.0%	2.1%
	5-high	2.1%	1.4%	2.3%
	5–1	1.4%	1.2%	1.5%
	t-stat.	4.75	4.56	4.45

#### Table XI

#### Average PVOL-CVOL of two-way sorted portfolios.

The sample between January 1996 and August 2015 includes 156,125 pairs of call and put options with the same ATM moneyness. Each month option observations are first sorted into quintiles based on the Amihud illiquidity measure. Within these quintiles, options are sorted into quintiles based on the difference between the call and put implied volatility. This table shows the average difference between call and put implied volatility for the different categories. Associated t-statistics are corrected for autocorrelation following Newey and West (1987).

			Put impli	ed VOL -	Call impl	lied VOL			mean	sd
		1-low	2	3	4	5-high	5-1	t-stat.	a	11
	1-low	-1.7%	-0.3%	0.3%	1.0%	2.8%	4.5%	14.38	0.4%	1.9%
pni	2	-2.2%	-0.3%	0.5%	1.4%	4.0%	6.1%	14.43	0.7%	2.6%
mih	3	-3.2%	-0.5%	0.6%	1.9%	5.5%	8.7%	16.04	0.9%	3.6%
$\mathbf{A}_1$	4	-4.3%	-0.7%	0.8%	2.4%	7.6%	11.9%	18.31	1.1%	4.9%
	5-high	-7.3%	-1.4%	1.0%	3.7%	11.9%	19.2%	21.81	1.6%	7.9%
	5–1	-5.6%	-1.1%	0.7%	2.7%	9.2%	14.7%		1.2%	6.0%
	t-stat.	-14.70	-7.08	9.40	22.38	27.17	23.50		10.46	23.64

#### Table XII

#### Average monthly synthetic future returns.

The sample between January 1996 and August 2015 includes 156,125 pairs of call and put options with the same ATM moneyness. Each month, option observations are first sorted into quintiles based on the Amihud liquidity measure. Within these quintiles, options are sorted into quintiles based on the difference between the call and put implied volatility. This table shows the average monthly returns of the portfolios for the different categories. The portfolio returns use an equal weighting of the returns of the long call, short put and short stock portfolios falling in the category. For the return calculation, the average of the closing bid and ask quotes is the reference beginning price. Associated t-statistics are corrected for autocorrelation following Newey and West (1987).

			Put impli	ied VOL -	Call impl	lied VOL			i	mean	sd
		1-low	2	3	4	5-high	5-1	t-stat.		8	ıll
	1-low	0.00%	0.01%	0.03%	0.05%	0.13%	0.12%	9.07		0.0%	0.2%
pnı	2	-0.01%	0.03%	0.05%	0.08%	0.22%	0.23%	8.80		0.1%	0.3%
mih	3	-0.02%	0.03%	0.06%	0.12%	0.31%	0.33%	9.66		0.1%	0.4%
A	4	-0.03%	0.03%	0.08%	0.14%	0.43%	0.46%	12.57		0.1%	0.5%
	5-high	-0.10%	0.03%	0.08%	0.18%	0.62%	0.71%	13.30		0.2%	0.8%
	5-1	-0.10%	0.02%	0.06%	0.13%	0.49%	0.59%			0.1%	0.6%
	t-stat.	-3.10	1.32	6.54	13.20	18.52	13.77			11.15	24.13

## Figure 1. Distribution of end user net demand across underlyings.

This figure plots the number of underlyings in net demand intervals. Net demand is defined as the difference between end user-initiated buys and sells per underlying. The sample comprises option trading data for 1,620 underlyings, traded on July 7, 2006, at the CBOE.



## Figure 2. Average option returns for Amihud deciles.

The 3–1 return is calculated as in Table III, but with a decile sorting on the Amihud measure and a second sorting on HV–IV into three portfolios. The average return for the Amihud decile is the equally weighted return of all delta-hedged calls or straddles in the decile.



#### Figure 3. Average empirical and simulated delta-hedged call returns.

The 3–1 empirical returns are calculated as in Table III, but with a decile sorting on the Amihud measure and a threequantile second sort on HV–IV. The 3–1 simulated returns are calculated as in Table V, but with a tercile sorting on HV–IV. Proportional transaction cost assumptions are k/2 = 0.05%, 0.1%, 0.15%, 0.2%, 0.25%, 0.3%, 0.35%, 0.4%, 0.45%, and 0.5%.



## REFERENCES

Amihud, Yakov, 2002, Illiquidity and stock returns: Cross-section and time-series effects, *Journal of Financial Markets* 5, 31–56.

An, Byeong-Je, Andrew Ang, Turan G. Bali, and Nusret Cakici, 2014, The joint cross section of stocks and options, *Journal of Finance* 69, 2279–2337.

Bali, Turan G., and Armen Hovakimian, 2009, Volatility spreads and expected stock returns, *Management Science* 55, 1797–1812.

Battalio, Robert, Brian Hatch, and Robert Jennings, 2004, Toward a national market system for US exchange–listed equity options, *Journal of Finance* 59, 933–962.

Bessembinder, Hendrik, 2003, Issues in assessing trade execution costs, *Journal of Financial Markets* 6, 233–257.

Black, Fisher, and Myron S. Scholes, 1973, The pricing of options and corporate liabilities, *Journal of Political Economy* 81, 637–654.

Bollen, Nicolas P. B., and Robert E. Whaley, 2004, Does net buying pressure affect the shape of implied volatility functions? *Journal of Finance* 59, 711–753.

Boyle, Phelim P., and Ton Vorst, 1992, Option replication in discrete time with transaction costs, *Journal of Finance* 47, 271–293.

Buraschi, Andrea, Fabio Trojani, and Andrea Vedolin, 2014, When uncertainty blows in the orchard: Comovement and equilibrium volatility risk premia, *Journal of Finance* 69, 101–137.

Cao, Jie, and Bing Han, 2013, Cross section of option returns and idiosyncratic stock volatility, *Journal of Financial Economics* 108, 231–249.

Carhart, Mark M., 1997, On persistence in mutual fund performance, *Journal of Finance* 52, 57–82.

Carr, Peter, and Liuren Wu, 2009, Variance risk premiums, *Review of Financial Studies* 22, 1311–1341.

Cetin, Umut, Robert Jarrow, Philip Protter, and Mitch Warachka, 2006, Pricing options in an extended Black Scholes economy with illiquidity: Theory and empirical evidence, *Review of Financial Studies* 19, 493–529.

Choy, Siu-Kai, and Jason Wei, 2016, Liquidity risk and expected option returns, Working paper, Shanghai University and University of Toronto.

Christoffersen, Peter, Ruslan Goyenko, Kris Jacobs, and Mehdi Karoui, 2015, Illiquidity premia in the equity options market, Working paper, McGill University, University of Houston, and University of Toronto.

Copeland, Thomas E., and Dan Galai, 1983, Information effects on the bid-ask spread, *Journal of Finance* 38, 1457–1469.

Corwin, Shane A., and Paul Schultz, 2012, A simple way to estimate bid–ask spreads from daily high and low prices, *Journal of Finance* 67, 719–760.

Coval, Joshua D., and Tyler Shumway, 2001, Expected option returns, *Journal of Finance* 56, 983–1009.

Driessen, Joost, Pascal J. Maenhout, and Grigory Vilkov, 2009, The price of correlation risk: Evidence from equity options, *Journal of Finance* 64, 1377–1406.

Easley, David, Soeren Hvidkjaer, and Maureen O'Hara, 2002, Is information risk a determinant of asset returns? *Journal of Finance* 57, 2185–2221.

Easley, David, and Maureen O'Hara, 1987, Price, trade size, and information in securities markets, *Journal of Financial Economics* 19, 69–90.

Easley, David, Maureen O'Hara, and Pulle Subrahmanya Srinivas, 1998, Option volume and stock prices: Evidence on where informed traders trade, *Journal of Finance* 53, 431–465.

Engle, Robert, and Breno Neri, 2010, The impact of hedging costs on the bid and ask spread in the options market, Working paper, New York University.

Fama, Eugene F., and Kenneth R. French, 1993, Common risk factors in the returns on stocks and bonds, *Journal of Financial Economics* 33, 3–56.

Frazzini, Andrea, and Lasse H. Pedersen, 2012, Embedded leverage, Working paper, New York University.

Gârleanu, Nicolae, Lasse Heje Pedersen, and Allen M. Poteshman, 2009, Demand-based option pricing, *Review of Financial Studies* 22, 4259–4299.

Glosten, Lawrence R., and Paul R. Milgrom, 1985, Bid, ask and transaction prices in a specialist market with heterogeneously informed traders, *Journal of Financial Economics* 14, 71–100.

Goyal, Amit, and Alessio Saretto, 2009, Cross-section of option returns and volatility, *Journal of Financial Economics* 94, 310–326.

Goyenko, Ruslan Y., Craig W. Holden, and Charles A. Trzcinka, 2009, Do liquidity measures measure liquidity? *Journal of Financial Economics* 92, 153–181.

Goyenko, Ruslan Y., Chayawat Ornthanalai, and Shengzhe Tang, 2015, Options illiquidity: Determinants and implications for stock returns, Working paper, McGill University and University of Toronto.

Hull, John, and Alan White, 1987, The pricing of options on assets with stochastic volatilities, *Journal of Finance* 42, 281–300.

Karakaya, Mete, 2014, Characteristics and expected returns in individual equity options, Working paper, University of Chicago.

Lakonishok, Josef, Inmoo Lee, Neil D. Pearson, and Allen M. Poteshman, 2007, Option market activity, *Review of Financial Studies* 20, 813–857.

Leland, Hayne E., 1985, Option pricing and replication with transactions costs, *Journal of Finance* 40, 1283–1301.

Muravyev, Dmitriy, 2016, Order flow and expected option returns, forthcoming, *Journal of Finance*.

Newey, Whitney K., and Kenneth D. West, 1987, A simple, positive semi-definite, heteroskedasticity and autocorrelation consistent covariance matrix, *Econometrica* 50, 703–708.

Ni, Sophie X., Jun Pan, and Allen M. Poteshman, 2008, Volatility information trading in the option market, *Journal of Finance* 63, 1059–1091.

Pan, Jun, and Allen M. Poteshman, 2006, The information in option volume for future stock prices, *Review of Financial Studies* 19, 871–908.

Roll, Richard, 1984, A simple implicit measure of the effective bid-ask spread in an efficient market, *Journal of Finance* 39, 1127–1139.

Schürhoff, Norman, and Alexandre Ziegler, 2011, Variance risk, financial intermediation, and the cross-section of expected option returns, Working paper, University of Lausanne and University of Zurich.

## **Appendix A: Liquidity Measure Calculations**

*Amihud measure:* Following Cao and Han (2013), we calculate the Amihud (2002) illiquidity measure for the month preceding the trading initiation date t as

$$ILLIQ_{i,t} = 1/m_{i,t} \sum_{d=t-m_{i,t}}^{t-1} \frac{|R_{i,d}|}{VOLD_{i,d}},$$

where  $m_{i,t}$  is the number of trading days from last month's trading initiation date until t - 1 with available return and volume data for stock i. The absolute daily total return  $|R_{i,d}|$  for stock i on day d is divided by the dollar trading volume  $VOLD_{i,d}$ , which we calculate by multiplying the closing price for stock i on day d with the trading volume on that date.

*Roll measure:* Roll (1984) introduced an estimator of bid–ask spreads based on the serial covariance of price changes. While changes of the fundamental stock value are assumed to be serially uncorrelated, closing prices are either bid or ask prices, which introduces negative serial correlation. We calculate the Roll spreads  $sp_{i,t}^{R}$  as

$$sp_{i,t}^R = 2\sqrt{-Cov_{i,t}}$$
 ,

where  $Cov_{i,t}$  is the covariance of the daily returns of the close prices for stock *i* in the month preceding the trading initiation date *t*. This is an estimate of the relative bid–ask spread. We drop observations with positive covariance values.

*Corwin–Schultz measure:* Corwin and Schultz (2012) show that bid–ask spreads can be estimated from daily high and low prices. Since daily high (low) prices are almost always buy (sell) trades, the ratio of these prices reflects the fundamental stock volatility and its bid–ask spread. The suggested bid–ask spread estimator uses the fact that the fundamental volatility increases proportionally with the length of the observation interval while the bid–ask spread does not. For every overlapping two-day period d, d + 1 within the month preceding the trading initiation date t, we calculate

$$\beta = \left[ \ln\left(\frac{H_d^{i,t}}{L_d^{i,t}}\right) \right]^2 + \left[ \ln\left(\frac{H_{d+1}^{i,t}}{L_{d+1}^{i,t}}\right) \right]^2, \qquad \gamma = \left[ \ln\left(\frac{H_{d,d+1}^{i,t}}{L_{d,d+1}^{i,t}}\right) \right]^2,$$

where  $H_d^{i,t}$  and  $L_d^{i,t}$  are the high and low prices, respectively, for stock *i* on day *d* and  $H_{d,d+1}^{i,t}$  and  $L_{d,d+1}^{i,t}$  are, respectively, the high and low prices in the two-day period. The bid–ask spread estimate for the two-day period can then be calculated as

$$sp_{i,t,d}^{CS} = \frac{2(e^{\alpha}-1)}{1+e^{\alpha}},$$

where

$$\alpha = \frac{\sqrt{2\beta} - \sqrt{\beta}}{3 - 2\sqrt{2}} - \sqrt{\frac{\gamma}{3 - 2\sqrt{2}}}.$$

The estimate of the relative spread for stock *i* on the trading initiation date *t* is the average of the two-day spread estimates for the preceding month.

*Size and trading volume:* We also use the underlying's market capitalization and trading volume as illiquidity measures. We calculate the log of market capitalization (size), where market capitalization is the number of shares outstanding times the underlying closing price the day preceding the trading initiation date. A stock's dollar trading volume is the number of shares traded on all U.S. exchanges the day preceding the trading initiation date multiplied by the closing prices. Shares outstanding, trading volumes, and the closing prices are from the OptionMetrics database.

## **Appendix B: Risk Factor Calculations**

*Fama–French and Carhart factors:* Since the returns of delta-hedged calls and straddles could still be exposed to stock price risk, we consider the Fama–French (1993) and Carhart (1997) factors as potential explanatory variables. These factors are calculated from the daily factor returns from Kenneth French's website. Since our monthly holding period starts at the beginning of the fourth week of the month and ends at the end of the third week, we do not use standard monthly returns. Instead, we compound the factor returns over the holding period from the trading initiation date until the trading day prior to expiration, at which point we also calculate the option payoffs. We calculate the factor return for one month as

$$\mathbf{F}_{t,t+\tau} = \prod_{d=1}^{N} (1+f_d),$$

where *N* is the number of trading days between *t* and  $t + \tau$  and  $f_d$  is the daily factor return of the *MKT* – *Rf*, *SMB*, *HML*, and *MOM* portfolios.

Zero-beta straddles: We use index and index component straddle returns to control for market volatility risk and common individual stock variance risk. We form zero-beta straddles similar to those of Coval and Shumway (2001). The zero-beta straddles are constructed the same day we initiate our trading strategy with one ATM call and one ATM put on the underlying. Call returns  $r_{C,i,t}$  and put returns  $r_{P,i,t}$ , referring to the underlying stock or index *i*, are calculated with the option payoffs at  $t + \tau$  and the option mid prices at *t* as the reference beginning price. These returns are then weighted so that the portfolio beta equals zero, leading to the zero-beta straddle return  $r_{zb,i,t}$ .

## **Appendix C: GARCH Calculations**

If available, we use five years of daily return data for the estimation of the GARCH(1,1) parameters. We drop a stock from the GARCH estimation if less than one month of return data are available, if five consecutive trading days have no return data, or if more than 10% of the returns are zero. We employ a maximum likelihood estimation for the GARCH(1,1) equation on the portfolio formation date t - 1:

$$\sigma_{i,t-1,d}^2 = \omega_{i,t-1} + \alpha_{i,t-1} u_{i,t-1,d-1}^2 + \beta_{i,t-1} \sigma_{i,t-1,d-1}^2,$$

where  $\omega_{i,t-1}$  is the product of the parameter  $\gamma_{i,t-1}$  and the long-term variance  $V_{i,t-1}$  for stock *i*,  $\sigma_{i,t-1,d-1}^2$  is the estimated variance for stock *i* on day *d* within the estimation period, and  $u_{i,t-1,d-1}^2$  is the squared stock return of the previous trading day. The weights  $\gamma_{i,t-1}$ ,  $\alpha_{i,t-1}$ , and  $\beta_{i,t-1}$  have to satisfy  $\gamma_{i,t-1} + \alpha_{i,t-1} + \beta_{i,t-1} = 1$ . Once  $\omega_{i,t-1}$ ,  $\alpha_{i,t-1}$ , and  $\beta_{i,t-1}$  are estimated, the long-term variance  $V_{i,t-1}$  can be deduced from this condition. Hull and White (1987) have suggested using the average variance rate during the life of the option when volatility is stochastic but uncorrelated with the asset price. We use the GARCH(1,1) model to forecast the volatility on the days between trading initiation *t* until maturity  $t + \tau$ . The average of these forecasts can be calculated with

$$\sigma(t+\tau)_{i}^{2} = 252 \left( V_{i,t-1} + \frac{1-e^{\tau \cdot \ln\left(\alpha_{i,t-1}+\beta_{i,t-1}\right)}}{-\ln\left(\alpha_{i,t-1}+\beta_{i,t-1}\right)\cdot\tau} \left[\sigma_{i,t-1,t-1}^{2} - V_{i,t-1}\right] \right),$$

assuming 252 trading days per year.

CFR working paper series



CFR Working Papers are available for download from www.cfr-cologne.de.

Hardcopies can be ordered from: Centre for Financial Research (CFR), Albertus Magnus Platz, 50923 Koeln, Germany.

No.	Author(s)	Title
16-08	S. Kanne, O. Korn, M.Uhrig-Homburg	Stock Illiquidity, Option Prices and Option Returns
16-07	S. Jaspersen	Market Power in the Portfolio: Product Market Competition and Mutual Fund Performance
16-06	O. Korn, MO. Rieger	Hedging With Regret
16-05	E. Theissen, C. Westheide	Call of Duty: Designated Market Maker Participation in Call Auctions
16-04	P. Gomber, S. Sagade, E. Theissen, M.C. Weber, C. Westheide	Spoilt for Choice: Order Routing Decisions in Fragmented Equity Markets
16-03	T.Martin, F. Sonnenburg	Managerial Ownership Changes and Mutual Fund Performance
16-02	A.Gargano, A. G. Rossi, R. Wermers	The Freedom of Information Act and the Race Towards Information Acquisition
16-01	G. Cici, S. Gibson, C. Rosenfeld	Cross-Company Effects of Common Ownership: Dealings Between Borrowers and Lenders With a Common Blockholder
2015		

No.	Author(s)	Title
15-17	O. Korn, L. Kuntz	Low-Beta Investment Strategies
15-16	D. Blake, A.G. Rossi, A. Timmermann, I. Tonks, R. Wermers	Network Centrality and Pension Fund Performance
15-15	S. Jank, E. Smajbegovic	Dissecting Short-Sale Performance: Evidence from Large Position Disclosures
15-14	M. Doumet, P. Limbach, E. Theissen	Ich bin dann mal weg: Werteffekte von Delistings deutscher Aktiengesellschaften nach dem Frosta-Urteil
15-13	G. Borisova, P.K. Yadav	Government Ownership, Informed Trading and Private Information

15-12	V. Agarwal, G.O. Aragon, Z. Shi	Funding Liquidity Risk of Funds of Hedge Funds: Evidence from their Holdings
15-11	L. Ederington, W. Guan, P.K. Yadav	Dealer spreads in the corporate Bond Market: Agent vs. Market-Making Roles
15-10	J.R. Black, D. Stock, P.K. Yadav	The Pricing of Different Dimensions of Liquidity: Evidence from Government Guaranteed Bank Bonds
15-09	V. Agarwal, H. Zhao	Interfund lending in mutual fund families: Role of internal capital markets
15-08	V. Agarwal, T. C. Green, H. Ren	Alpha or Beta in the Eye of the Beholder: What drives Hedge Fund Flows?
15-07	V. Agarwal, S. Ruenzi, F. Weigert	Tail risk in hedge funds: A unique view from portfolio holdings
15-06	C. Lan, F. Moneta, R. Wermers	Mutual Fund Investment Horizon and Performance
15-05	L.K. Dahm, C. Sorhage	Milk or Wine: Mutual Funds' (Dis)economies of Life
15-04	A. Kempf, D. Mayston, M. Gehde-Trapp, P. K. Yadav	Resiliency: A Dynamic View of Liquidity
15-03	V. Agarwal, Y. E. Arisoy, N. Y. Naik	Volatility of Aggregate Volatility and Hedge Funds Returns
15-02	G. Cici, S. Jaspersen, A. Kempf	Speed of Information Diffusion within Fund Families
15-01	M. Baltzer, S. Jank, E. Smajlbegovic	Who trades on momentum?

No.	Author(s)	Title
14-15	M. Baltzer, S. Jank, E.Smajlbegovic	Who Trades on Monumentum?
14-14	G. Cici, L. K. Dahm, A. Kempf	Trading Efficiency of Fund Families: Impact on Fund Performance and Investment Behavior
14-13	V. Agarwal, Y. Lu, S. Ray	Under one roof: A study of simultaneously managed hedge funds and funds of hedge funds
14-12	P. Limbach, F. Sonnenburg	Does CEO Fitness Matter?
14-11	G. Cici, M. Gehde-Trapp, M. Göricke, A. Kempf	What They Did in their Previous Life: The Investment Value of Mutual Fund Managers' Experience outside the Financial Sector
14-10	O. Korn, P. Krischak, E. Theissen	Illiquidity Transmission from Spot to Futures Markets
14-09	E. Theissen, L. S. Zehnder	Estimation of Trading Costs: Trade Indicator Models Revisited
14-08	C. Fink, E. Theissen	Dividend Taxation and DAX Futures Prices

14-07	F. Brinkmann, O. Korn	Risk-adjusted Option-implied Moments
14-06	J. Grammig, J. Sönksen	Consumption-Based Asset Pricing with Rare Disaster Risk
14-05	J. Grammig, E. Schaub	Give me strong moments and time – Combining GMM and SMM to estimate long-run risk asset pricing
14-04	C. Sorhage	Outsourcing of Mutual Funds' Non-core Competencies
14-03	D. Hess, P. Immenkötter	How Much Is Too Much? Debt Capacity And Financial Flexibility
14-02	C. Andres, M. Doumet, E. Fernau, E. Theissen	The Lintner model revisited: Dividends versus total payouts

14-01 N.F. Carline, S. C. Linn, P. Corporate Governance and the Nature of Takeover Resistance K. Yadav

## 2013

No.	Author(s)	Title
13-11	R. Baule, O. Korn, S. Saßning	Which Beta is Best? On the Information Content of Option-implied Betas
13-10	V. Agarwal, L. Ma, K. Mullally	Managerial Multitasking in the Mutual Fund Industry
13-09	M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers	Seasonal Asset Allocation: Evidence from Mutual Fund Flows
13-08	F. Brinkmann, A. Kempf, O. Korn	Forward-Looking Measures of Higher-Order Dependencies with an Application to Portfolio Selection
13-07	G. Cici, S. Gibson, Y. Gunduz, J.J. Merrick, Jr.	Market Transparency and the Marking Precision of Bond Mutual Fund Managers
13-06	S. Bethke, M. Gehde- Trapp, A. Kempf	Investor Sentiment, Flight-to-Quality, and Corporate Bond Comovement
13-05	P. Schuster, M. Trapp, M. Uhrig-Homburg	A Heterogeneous Agents Equilibrium Model for the Term Structure of Bond Market Liquidity
13-04	V. Agarwal, K. Mullally, Y. Tang, B. Yang	Mandatory Portfolio Disclosure, Stock Liquidity, and Mutual Fund Performance
13-03	V. Agarwal, V. Nanda, S.Ray	Institutional Investment and Intermediation in the Hedge Fund Industry
13-02	C. Andres, A. Betzer, M. Doumet, E. Theissen	Open Market Share Repurchases in Germany: A Conditional Event Study Approach
13-01	J. Gaul, E. Theissen	A Partially Linear Approach to Modelling the Dynamics of Spot and Futures Price

## 2012

No.	Author(s)	Title
12-12	M. Gehde-Trapp, Y. Gündüz, J. Nasev	The liquidity premium in CDS transaction prices: Do frictions matter?

\_\_\_\_

12-11	Y. Wu, R. Wermers, J. Zechner	Governance and Shareholder Value in Delegated Portfolio Management: The Case of Closed-End Funds
12-10	M. Trapp, C. Wewel	Transatlantic Systemic Risk
12-09	G. Cici, A. Kempf, C. Sorhage	Do Financial Advisors Provide Tangible Benefits for Investors? Evidence from Tax-Motivated Mutual Fund Flows
12-08	S. Jank	Changes in the composition of publicly traded firms: Implications for the dividend-price ratio and return predictability
12-07	G. Cici, C. Rosenfeld	A Study of Analyst-Run Mutual Funds: The Abilities and Roles of Buy-Side Analysts
12-06	A. Kempf, A. Pütz, F. Sonnenburg	Fund Manager Duality: Impact on Performance and Investment Behavior
12-05	L. Schmidt, A. Timmermann, R. Wermers	Runs on Money Market Mutual Funds
12-04	R. Wermers	A matter of style: The causes and consequences of style drift in institutional portfolios
12-02	C. Andres, E. Fernau, E. Theissen	Should I Stay or Should I Go? Former CEOs as Monitors
12-01	L. Andreu, A. Pütz	Choosing two business degrees versus choosing one: What does it tell about mutual fund managers' investment behavior?

No.	Author(s)	Title
11-16	V. Agarwal, JP. Gómez, R. Priestley	Management Compensation and Market Timing under Portfolio Constraints
11-15	T. Dimpfl, S. Jank	Can Internet Search Queries Help to Predict Stock Market Volatility?
11-14	P. Gomber, U. Schweickert, E. Theissen	Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach
11-13	D. Hess, S. Orbe	Irrationality or Efficiency of Macroeconomic Survey Forecasts? Implications from the Anchoring Bias Test
11-12	D. Hess, P. Immenkötter	Optimal Leverage, its Benefits, and the Business Cycle
11-11	N. Heinrichs, D. Hess, C. Homburg, M. Lorenz, S. Sievers	Extended Dividend, Cash Flow and Residual Income Valuation Models – Accounting for Deviations from Ideal Conditions
11-10	A. Kempf, O. Korn, S. Saßning	Portfolio Optimization using Forward - Looking Information
11-09	V. Agarwal, S. Ray	Determinants and Implications of Fee Changes in the Hedge Fund Industry
11-08	G. Cici, LF. Palacios	On the Use of Options by Mutual Funds: Do They Know What They Are Doing?
11-07	V. Agarwal, G. D. Gay, L. Ling	Performance inconsistency in mutual funds: An investigation of window-dressing behavior
11-06	N. Hautsch, D. Hess, D. Veredas	The Impact of Macroeconomic News on Quote Adjustments, Noise, and Informational Volatility

11-05	G. Cici	The Prevalence of the Disposition Effect in Mutual Funds' Trades
11-04	S. Jank	Mutual Fund Flows, Expected Returns and the Real Economy
11-03	G.Fellner, E.Theissen	Short Sale Constraints, Divergence of Opinion and Asset Value: Evidence from the Laboratory
11-02	S.Jank	Are There Disadvantaged Clienteles in Mutual Funds?
11-01	V. Agarwal, C. Meneghetti	The Role of Hedge Funds as Primary Lenders

No.	Author(s)	Title
10-20	G. Cici, S. Gibson, J.J. Merrick Jr.	Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds
10-19	J. Hengelbrock, E. Theissen, C. Westheide	Market Response to Investor Sentiment
10-18	G. Cici, S. Gibson	The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings
10-17	D. Hess, D. Kreutzmann, O. Pucker	Projected Earnings Accuracy and the Profitability of Stock Recommendations
10-16	S. Jank, M. Wedow	Sturm und Drang in Money Market Funds: When Money Market Funds Cease to Be Narrow
10-15	G. Cici, A. Kempf, A. Puetz	The Valuation of Hedge Funds' Equity Positions
10-14	J. Grammig, S. Jank	Creative Destruction and Asset Prices
10-13	S. Jank, M. Wedow	Purchase and Redemption Decisions of Mutual Fund Investors and the Role of Fund Families
10-12	S. Artmann, P. Finter, A. Kempf, S. Koch, E. Theissen	The Cross-Section of German Stock Returns: New Data and New Evidence
10-11	M. Chesney, A. Kempf	The Value of Tradeability
10-10	S. Frey, P. Herbst	The Influence of Buy-side Analysts on Mutual Fund Trading
10-09	V. Agarwal, W. Jiang, Y. Tang, B. Yang	Uncovering Hedge Fund Skill from the Portfolio Holdings They Hide
10-08	V. Agarwal, V. Fos, W. Jiang	Inferring Reporting Biases in Hedge Fund Databases from Hedge Fund Equity Holdings
10-07	V. Agarwal, G. Bakshi, J. Huij	Do Higher-Moment Equity Risks Explain Hedge Fund Returns?
10-06	J. Grammig, F. J. Peter	Tell-Tale Tails: A data driven approach to estimate unique market information shares
10-05	K. Drachter, A. Kempf	Höhe, Struktur und Determinanten der Managervergütung- Eine Analyse der Fondsbranche in Deutschland
10-04	J. Fang, A. Kempf, M. Trapp	Fund Manager Allocation

10-03	P. Finter, A. Niessen- Ruenzi, S. Ruenzi	The Impact of Investor Sentiment on the German Stock Market
10-02	D. Hunter, E. Kandel, S. Kandel, R. Wermers	Mutual Fund Performance Evaluation with Active Peer Benchmarks
10-01	S. Artmann, P. Finter, A. Kempf	Determinants of Expected Stock Returns: Large Sample Evidence from the German Market

No.	Author(s)	Title
09-17	E. Theissen	Price Discovery in Spot and Futures Markets: A Reconsideration
09-16	М. Тгарр	Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-15	A. Betzer, J. Gider, D.Metzger, E. Theissen	Strategic Trading and Trade Reporting by Corporate Insiders
09-14	A. Kempf, O. Korn, M. Uhrig-Homburg	The Term Structure of Illiquidity Premia
09-13	W. Bühler, M. Trapp	Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets
09-12	W. Bühler, M. Trapp	Explaining the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-11	S. J. Taylor, P. K. Yadav, Y. Zhang	Cross-sectional analysis of risk-neutral skewness
09-10	A. Kempf, C. Merkle, A. Niessen-Ruenzi	Low Risk and High Return – Affective Attitudes and Stock Market Expectations
09-09	V. Fotak, V. Raman, P. K. Yadav	Naked Short Selling: The Emperor's New Clothes?
09-08	F. Bardong, S.M. Bartram, P.K. Yadav	Informed Trading, Information Asymmetry and Pricing of Information Risk: Empirical Evidence from the NYSE
09-07	S. J. Taylor , P. K. Yadav, Y. Zhang	The information content of implied volatilities and model-free volatility expectations: Evidence from options written on individual stocks
09-06	S. Frey, P. Sandas	The Impact of Iceberg Orders in Limit Order Books
09-05	H. Beltran-Lopez, P. Giot, J. Grammig	Commonalities in the Order Book
09-04	J. Fang, S. Ruenzi	Rapid Trading bei deutschen Aktienfonds: Evidenz aus einer großen deutschen Fondsgesellschaft
09-03	A. Banegas, B. Gillen, A. Timmermann, R. Wermers	The Cross-Section of Conditional Mutual Fund Performance in European Stock Markets
09-02	J. Grammig, A. Schrimpf, M. Schuppli	Long-Horizon Consumption Risk and the Cross-Section of Returns: New Tests and International Evidence
09-01	O. Korn, P. Koziol	The Term Structure of Currency Hedge Ratios

No.	Author(s)	Title
08-12	U. Bonenkamp, C. Homburg, A. Kempf	Fundamental Information in Technical Trading Strategies
08-11	O. Korn	Risk Management with Default-risky Forwards
08-10	J. Grammig, F.J. Peter	International Price Discovery in the Presence of Market Microstructure Effects
08-09	C. M. Kuhnen, A. Niessen	Public Opinion and Executive Compensation
08-08	A. Pütz, S. Ruenzi	Overconfidence among Professional Investors: Evidence from Mutual Fund Managers
08-07	P. Osthoff	What matters to SRI investors?
08-06	A. Betzer, E. Theissen	Sooner Or Later: Delays in Trade Reporting by Corporate Insiders
08-05	P. Linge, E. Theissen	Determinanten der Aktionärspräsenz auf Hauptversammlungen deutscher Aktiengesellschaften
08-04	N. Hautsch, D. Hess, C. Müller	Price Adjustment to News with Uncertain Precision
08-03	D. Hess, H. Huang, A. Niessen	How Do Commodity Futures Respond to Macroeconomic News?
08-02	R. Chakrabarti, W. Megginson, P. Yadav	Corporate Governance in India
08-01	C. Andres, E. Theissen	Setting a Fox to Keep the Geese - Does the Comply-or-Explain Principle Work?

No.	Author(s)	Title
07-16	M. Bär, A. Niessen, S. Ruenzi	The Impact of Work Group Diversity on Performance: Large Sample Evidence from the Mutual Fund Industry
07-15	A. Niessen, S. Ruenzi	Political Connectedness and Firm Performance: Evidence From Germany
07-14	O. Korn	Hedging Price Risk when Payment Dates are Uncertain
07-13	A.Kempf, P. Osthoff	SRI Funds: Nomen est Omen
07-12	J. Grammig, E. Theissen, O. Wuensche	Time and Price Impact of a Trade: A Structural Approach
07-11	V. Agarwal, J. R. Kale	On the Relative Performance of Multi-Strategy and Funds of Hedge Funds
07-10	M. Kasch-Haroutounian, E. Theissen	Competition Between Exchanges: Euronext versus Xetra
07-09	V. Agarwal, N. D. Daniel, N. Y. Naik	Do hedge funds manage their reported returns?
07-08	N. C. Brown, K. D. Wei, R. Wermers	Analyst Recommendations, Mutual Fund Herding, and Overreaction in Stock Prices
07-07	A. Betzer, E. Theissen	Insider Trading and Corporate Governance: The Case of Germany

07-06	V. Agarwal, L. Wang	Transaction Costs and Value Premium
07-05	J. Grammig, A. Schrimpf	Asset Pricing with a Reference Level of Consumption: New Evidence from the Cross-Section of Stock Returns
07-04	V. Agarwal, N.M. Boyson, N.Y. Naik	Hedge Funds for retail investors? An examination of hedged mutual funds
07-03	D. Hess, A. Niessen	The Early News Catches the Attention: On the Relative Price Impact of Similar Economic Indicators
07-02	A. Kempf, S. Ruenzi, T. Thiele	Employment Risk, Compensation Incentives and Managerial Risk Taking - Evidence from the Mutual Fund Industry -
07-01	M. Hagemeister, A. Kempf	CAPM und erwartete Renditen: Eine Untersuchung auf Basis der Erwartung von Marktteilnehmern
2006		
No.	Author(s)	Title
06-13	S. Čeljo-Hörhager, A. Niessen	How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study
06-12	R. Wermers, Y. Wu, J. Zechner	Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers
06-11	U. v. Lilienfeld-Toal, S. Ruenzi	Why Managers Hold Shares of Their Firm: An Empirical Analysis
06-10	A. Kempf, P. Osthoff	The Effect of Socially Responsible Investing on Portfolio Performance
06-09	R. Wermers, T. Yao, J. Zhao	Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach
06-08	M. Hoffmann, B. Kempa	The Poole Analysis in the New Open Economy Macroeconomic Framework
06-07	K. Drachter, A. Kempf, M. Wagner	Decision Processes in German Mutual Fund Companies: Evidence from a Telephone Survey
06-06	J.P. Krahnen, F.A. Schmid, E. Theissen	Investment Performance and Market Share: A Study of the German Mutual Fund Industry
06-05	S. Ber, S. Ruenzi	On the Usability of Synthetic Measures of Mutual Fund Net- Flows
06-04	A. Kempf, D. Mayston	Liquidity Commonality Beyond Best Prices
06-03	O. Korn, C. Koziol	Bond Portfolio Optimization: A Risk-Return Approach
06-02	O. Scaillet, L. Barras, R. Wermers	False Discoveries in Mutual Fund Performance: Measuring Luck in Estimated Alphas
06-01	A. Niessen, S. Ruenzi	Sex Matters: Gender Differences in a Professional Setting

No.	Author(s)	Title
05-16	E. Theissen	An Analysis of Private Investors' Stock Market Return Forecasts
05-15	T. Foucault, S. Moinas, E. Theissen	Does Anonymity Matter in Electronic Limit Order Markets
05-14	R. Kosowski, A. Timmermann, R. Wermers, H. White	Can Mutual Fund "Stars" Really Pick Stocks? New Evidence from a Bootstrap Analysis

05-13	D. Avramov, R. Wermers	Investing in Mutual Funds when Returns are Predictable
05-12	K. Griese, A. Kempf	Liquiditätsdynamik am deutschen Aktienmarkt
05-11	S. Ber, A. Kempf, S. Ruenzi	Determinanten der Mittelzuflüsse bei deutschen Aktienfonds
05-10	M. Bär, A. Kempf, S. Ruenzi	Is a Team Different From the Sum of Its Parts? Evidence from Mutual Fund Managers
05-09	M. Hoffmann	Saving, Investment and the Net Foreign Asset Position
05-08	S. Ruenzi	Mutual Fund Growth in Standard and Specialist Market Segments
05-07	A. Kempf, S. Ruenzi	Status Quo Bias and the Number of Alternatives - An Empirical Illustration from the Mutual Fund Industry
05-06	J. Grammig, E. Theissen	Is Best Really Better? Internalization of Orders in an Open Limit Order Book
05-05	H. Beltran-Lopez, J. Grammig, A.J. Menkveld	Limit order books and trade informativeness
05-04	M. Hoffmann	Compensating Wages under different Exchange rate Regimes
05-03	M. Hoffmann	Fixed versus Flexible Exchange Rates: Evidence from Developing Countries
05-02	A. Kempf, C. Memmel	Estimating the Global Minimum Variance Portfolio
05-01	S. Frey, J. Grammig	Liquidity supply and adverse selection in a pure limit order book market

No.	Author(s)	Title
04-10	N. Hautsch, D. Hess	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery
04-09	A. Kempf, K. Kreuzberg	Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation
04-08	N.F. Carline, S.C. Linn, P.K. Yadav	Operating performance changes associated with corporate mergers and the role of corporate governance
04-07	J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav	Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze
04-06	N.Y. Naik, P.K. Yadav	Trading Costs of Public Investors with Obligatory and Voluntary Market-Making: Evidence from Market Reforms
04-05	A. Kempf, S. Ruenzi	Family Matters: Rankings Within Fund Families and Fund Inflows
04-04	V. Agarwal, N.D. Daniel, N.Y. Naik	Role of Managerial Incentives and Discretion in Hedge Fund Performance
04-03	V. Agarwal, W.H. Fung, J.C. Loon, N.Y. Naik	Risk and Return in Convertible Arbitrage: Evidence from the Convertible Bond Market
04-02	A. Kempf, S. Ruenzi	Tournaments in Mutual Fund Families
04-01	I. Chowdhury, M. Hoffmann, A. Schabert	Inflation Dynamics and the Cost Channel of Monetary Transmission

centre for Financial Research

cfr/university of cologne Albertus-Magnus-Platz D-50923 cologne Fon +49[0]221-470-6995 Fax +49[0]221-470-3992 Kempf@cfr-cologne.de WWW.cfr-cologne.de